Urbanization *without* Structural Transformation: Evidence from Consumption Cities in Africa^{*}

Rémi JEDWAB^a

^a Department of Economics, George Washington University

This Version: February 28th, 2013

Abstract: Africa has recently experienced dramatic urbanization. Standard theories of structural transformation cannot explain this result, as it was not driven by a green revolution or an industrial revolution, but by natural resource exports. I explain how the Engel curve implies that resource windfalls are disproportionately spent on urban goods and services, which gives rise to "consumption cities". I illustrate this theory using both cross-country evidence and within-country evidence from Ivory Coast and Ghana using new data spanning one century and two identification strategies (an instrumental variables strategy and a fixed effects approach). I find a strong causal effect of the production of cocoa, a ruralbased natural resource, on the growth of cities. I discuss the implications of urbanization *without* structural transformation for long-run growth.

Keywords: Urbanization; Structural Change; Resource Curse; Africa **JEL classification:** L16; N17; O18; O40; O55; R10

^{*}Remi Jedwab, Department of Economics, George Washington University (e-mail: jedwab@gwu.edu). I would like to thank Denis Cogneau, Robin Burgess, Guy Michaels, Douglas Gollin, Francisca Antman, David Atkin, Gareth Austin, Tania Barham, Timothy Besley, François Bourguignon, Paula Bustos, Francesco Caselli, Antonio Ciccone, Luc Christiansen, Uwe Deichmann, David Dorn, Alain de Janvry, Stefan Dercon, Marcel Fafchamps, Thibault Fally, James Fenske, James Foster, Nicola Gennaioli, Elise Huillery, Wolfgang Keller, Somik Lall, Sylvie Lambert, Pete Lanjouw, Stephan Litschig, Margaret McMillan, Rocco Macchiavello, Karen Macours, James Markusen, William Masters, Stelios Michalopoulos, Edward Miguel, Gianmarco Ottaviano, Henry Overman, Gerard Padró i Miquel, Torsten Persson, Thomas Piketty, Jean-Philippe Platteau, Giacomo Ponzetto, Diego Puga, Elizabeth Sadoulet, Harris Selod, Stephen Smith, Adam Storeygard, Alessandro Tarozzi, Francis Teal, Christopher Udry, Anthony Venables, Dietrich Vollrath, David Weil, Christopher Woodruff, and seminar audiences at PSE, LSE, Namur, Warwick, UPF-CREI, CEMFI, Paris (BREAD/CEPR/AMID Pre-Conference, EUDN Scientific Conference, ABCDE World Bank), Oxford (CSAE Seminar, CSAE Conference, Oxcarre Conference), London (African Economic History Workshop, AMID Summer School), Mombasa (IGC), Berkeley (PACDEV), Yale (NEUDC), World Bank FEU-DEC, George Washington University, Colorado-Boulder, Sussex, Central European University (GDN), Malaga (SAEe) and Tinbergen (EUDN) for very helpful comments. I thank the Ivorian National Institute for Statistics and the Ghanaian Statistical Service for giving me access to the surveys. I am grateful to Francois Moriconi-Ebrard from GEOPOLIS, Vincent Anchirinah and Frederick Amon-Armah from CRIG, Sandrine Mesple-Somps from DIAL-IRD and Elise Huillery from Science-Po for their help with data collection. I thank the Paris-Jourdan Sciences-Economiques UMR at the Paris School of Economics, the Suntory and Toyota International Centres for Economics and Related Disciplines at the London School of Economics, the Economic and Social Research Council [RES-167-25-0214] and the "Actors, Markets, and Institutions in Developing Countries" ITN at the Center for Economic Policy Research for financial assistance.

1. INTRODUCTION

Urbanization is typically seen as a consequence of economic growth. As a country develops, people move out of the rural-based agricultural sector into the urban-based manufacturing and service sectors. Standard structural transformation models identify "labor push" and "labor pull" factors as the main drivers of this transition. The labor push approach shows how a rise in agricultural productivity - a *green revolution* - reduces the "food problem" and releases labor for the modern sector. The labor pull approach describes how a rise in non-agricultural productivity - a *tracts* underemployed labor from agriculture into the modern sector .¹ These mechanisms all lead to greater non-agricultural employment, and thus greater urban employment as a proportion of total workforce.

However, in many parts of the world, urbanization has occurred without structural transformation towards manufacturing and services (Gollin, Jedwab & Vollrath, 2013). Figure 1 shows the relationship between urbanization and the fraction of economic activity engaged in manufacturing and services in 2000 for 119 developing countries, along with a quadratic fit line. There are many countries for which the processes of urbanization and structural transformation are disconnected. An explanation for this paradox can be seen by breaking up the sample based on the importance of natural resource exports.² Countries for which resource exports account for more than 10% of GDP are denoted in grey. These countries make up the vast majority of the countries that urbanized without structural transformation.

Sub-Saharan Africa offers a perfect example of this type of urbanization. At the turn of the 20th century, Africa's urbanization rate was 5%, the same level as in Medieval Europe (Bairoch, 1988). In 1960, Africa's urbanization rate was 15%, as low as in the European Renaissance. It is now around 40%, as high as contemporary Asia, or developed countries after the Industrial Revolution.³ Standard structural transformation models can not account for African urbanization. First, there has been no green revolution. Africa's food yields have remained low (Evenson & Gollin, 2003; Caselli, 2005; Restuccia, Yang & Zhu, 2008); in 2009, cereal yields were 2.8 times lower than in Asia, while yields were 2.1 times lower for starchy roots (FAO, 2012). Second, there has been no industrial revolution. Africa's manufacturing and service sectors are relatively small and unproductive (McMillan & Rodrik, 2011; Badiane, 2011; Jedwab & Darko Osei, 2012). In 2007, employment shares in industry and services were 10% and 26% for Africa, but 24% and 35% for Asia, and African labor productivity was 1.7 and 3.5 times lower in industry and services, respectively (World Bank, 2012). Third, while Asia has followed the standard pattern of urbanization and structural change, Africa's urbanization appears to be driven by a natural resource revolution (Gollin, Jedwab & Vollrath, 2013). This is confirmed by Figure 2 which plots the urbanization rate in 2000 against the average contribution of resource exports to GDP in 1960-2000 for Africa.⁴ This relationship

¹For studies that have focused on the link between a green revolution and structural change, see Schultz (1953), Caselli & Coleman II (2001), Gollin, Parente & Rogerson (2002, 2007), Nunn & Qian (2011) and Michaels, Rauch & Redding (2012). For studies of an industrial revolution, see Lewis (1954), Harris & Todaro (1970), Hansen & Prescott (2002), Lucas (2004, 2009) and Alvarez-Cuadrado & Poschke (2011). An industrial revolution facilitates the modernization of agriculture (Restuccia, Yang & Zhu, 2008; Yang & Zhu, 2010). Alternatively, a country with a comparative advantage in manufacturing can import food (Matsuyama, 1992; Yi & Zhang, 2011).

²My category of "natural resource exports" includes point source natural resources, such as oil and minerals, and non-point source natural resources, such plantation crops and timber.

³I alternatively use the expressions "Sub-Saharan Africa" and "Africa" in the rest of the paper to refer to 46 countries south of the Sahara. Africa's urbanization rate was 37% in 2010, while it was 40% for Asia (United Nations, 2011). It was 45% in England in 1850 and 42% in the U.S. in 1910 (Bairoch, 1988). I also use data on cities larger than 750,000 to find that they make up 39% of urban population in Africa and 44% in Asia.

⁴The historical contribution of natural resource exports to GDP is more than 10% for 34 countries. The most

cannot be explained by resource production industries being large employers of urban workers. Since point-source natural resources are capital-intensive, their numerical contribution to urban employment is small. For example, mining and quarrying only account for 0.9% of Africa's urban workforce in 2000.⁵ Cash crops mostly contribute to rural employment.

In this paper, I study the role of resource exports in creating a disconnection between the processes of urbanization and structural transformation. The first part of the paper establishes that the pattern seen for Africa (Figure 2) is robust to other factors of urbanization. A contribution of this study is to document that there are different paths of urbanization. I also investigate whether these different paths are associated with different sectoral compositions of the urban workforce, which is important for both economic theory and policy.

The second part of the paper offers a conceptual framework for thinking about the role of resource exports in urbanization. I examine different models in the literature and explain how they could be modified to account for the stylized facts described above. Given differences between the international price and local production costs, the export of natural resources has often generated a considerable surplus for producing countries. If the Engel curve implies that this surplus is mainly spent on urban goods and services, resource exports drive urbanization. Since greater opportunities in the urban sector attract people to cities, the theory is in line with the labor pull hypothesis, except the resource sector is the main driver of this transformation. First, as wealth is created in the non-urban sector and spent in the urban sector, these "consumption cities" arise due to demand factors. Second, if manufactured goods and tradable services are imported from abroad, the mix of urban workers is heavily skewed towards non-tradable services. These countries urbanize without structural transformation. Third, countries that specialize in the export of (urban-based) manufactured goods and tradable services urbanize with structural transformation. This specialization effect gives rise to "production cities", with a mix of workers in tradable and non-tradable sectors.

The third part of the paper provides within-country evidence consistent with the stylized facts and channels discussed above. I use new data from Ghana and Ivory Coast to investigate the causal effect of the production of cocoa, a rural-based natural resource, on the growth of cities at the district level over one century. Cocoa production boomed in the 1920's in Ghana and the 1950's in Ivory Coast (Ruf, 1995a), when they became the two largest producers in the world.⁶ Overall, cocoa production has accounted for 60% of their exports after 1948. I believe that examining cocoa in these two countries provides me with the perfect natural experiment to study the role of resource exports in urbanization. Given the scarcity of historical data for African countries, it is usually very difficult to study questions over such a long period. However, I have used various historical sources to recreate district-level data on resource production and urbanization in Ghana from 1901-2000 and Ivory Coast from 1948-1998. This would not have been possible for most other African countries. They are now two of the most urbanized countries in Africa, with urbanization rates around 50% in 2010; although, cocoa is produced on farms mostly owned by smallholders. Cocoa cultivation has few production linkages. Its mode of production has remained very traditional, with limited used of modern (urban) inputs. There is no local transformation of cocoa beans and transportation activities

urbanized countries export fuels and mineral products (Gabon, Republic of the Congo, South Africa, Botswana, Nigeria, Angola, Namibia, Liberia, Zambia) or plantation crops (Ivory Coast, Ghana, Senegal, Gambia, Cameroon).

⁵For example, Angola's urbanization rate was 15% before the oil boom in the 1970's, but it was 58% in 2010. Crude oil now accounts for 62% of GDP, but it employs fewer than 10,000 nationals. Botswana's urbanization rate is similar, but while the diamond sector accounts for 38% of GDP, it only provides employment for 13,000 people.

⁶Since Ivory Coast is a major coffee exporter and since Ivorian cocoa farmers also produce coffee that shares the same agronomic features as cocoa, I examine both. I alternatively use the expressions "cocoa production" and "cash crop production" in the rest of the paper, although they always include coffee for Ivory Coast.

generated by their export contribute little to urban employment. Only consumption linkages can account for this relationship. Lastly, cocoa cultivation has four unique features, which I exploit in order to identify its causal effect on the extent and the type of urban growth.

The first unique feature of cocoa production is that only forested areas are suitable for cocoa cultivation, which limits production to the South in each country. Cocoa is produced by "consuming" the forest (Ruf, 1995*a*,*b*), as cocoa smallholders go to a patch of virgin forest and replace forest trees with cocoa trees. The causality is unlikely to run from cities to cash crop production because settlement in the forest was historically limited due to tall trees, thick vegetation, high humidity and disease incidence. Farmers overcame these constraints when they achieved an income high enough to pay the fixed costs of deforestation, which was the case with cocoa. Given its traditional mode of production, there is also no role for cities in the diffusion of technological innovations used in this sector.

Within the forest, some areas are relatively more suitable for cocoa cultivation due to richer soil nutrients, and yields are much higher for the same level of inputs. Interestingly, highly suitable and poorly suitable areas are not different to the naked eye, since they only differ with the respect of the soil content. I estimate a long difference model: I regress the total district value of cash crop production on the change in the district urbanization rate between the beginning and the end of the period (1948 and 1998 for Ivory Coast, 1901 and 2000 for Ghana). I classify the districts into three groups: the *non-suitable*, the *poorly suitable* and the *highly suitable* districts. We expect the cash crop production over the long period to be much higher in the latter group, which then has a strong effect on urbanization. That is why I instrument the total district value of cash crop production with a dummy whose value is one if the district is highly suitable. I can also restrict the analysis to the suitable districts only. The instrument then captures the effect of being relatively more "suitable" than the other districts of the tropical forest, which makes it an even stronger identification strategy.

An unique agronomic feature of cocoa production is that yields begin to decline 20 years after planting (Ruf, 1995*a*,*b*). Cocoa trees become too old and must be replaced. However, the forest initially provides agronomic benefits (e.g., a high level of humidity and abundant rainfall) which are also extinguished after 20 years of local deforestation. Replanted cocoa trees die or are much less productive, which force farmers to move to a new forest a few years later, usually 25 years after cocoa trees were first planted. This extensive production strategy gives rise to district production cycles that can last several decades. For historical reasons, the cocoa front started in the East of each country. Production first spread out one century ago in the South-East of Ghana, in the vicinity of Aburi Botanical Gardens from where the British colonizer distributed cocoa seedlings to the local farmers. Production later started in the Southeast of Ivory Coast because the local tribes had heard of the wealth of the Ghanaian cocoa farmers and also wanted to adopt the crop. These factors resulted in the cocoa front moving across the South of each country from east to west. As production moved westward, new cities appeared in the West. Thus, when a district's production booms, we expect to see a rise in its urbanization rate. Therefore, I estimate a short difference model with district fixed effects: I regress the district value of cocoa production on the change in the district urbanization rate using several periods (1948, 1955, 1965, 1975, 1988, 1998 for Ivory Coast). I do not have panel data on district urbanization rates for Ghana (1901-2000), as administrative boundaries have been considerably modified across years, this impeding any consistent reaggregation. By using the panel dimension of the data and including district fixed effects, I capture the effects of these local resource booms on local urbanization. I also include district time trends and region-year fixed effects. The identification then comes from non-linear variations within districts over time, while simultaneously controlling for regional effects specific to some years. To see any

effect for a district, it must be that cocoa production booms, across periods and relatively to the neighboring districts of the same region, in a difference-in-difference spirit.

Finally, I use the westward wave to identify the short-term and long-term effects of resource exports on the sectoral composition of these consumption cities. I do not find that cities collapse in the old producing areas in the east. Therefore, I can compare cities in the new and old producing areas using cross-sectional employment data for the recent period. Although the cities in the east have boomed relatively earlier and have lost their access to the resource rent, I find that these cities are still not specialized at all in manufactured goods and tradable services. I also find that the employment share of these sectors is relatively small in the government districts, which I define here as the districts of the capital and second largest city. In the end, both the old producing and government districts have urbanized without structural transformation. Lastly, I use cross-sectional data on capital accumulation and infrastructure investments to investigate the mechanisms of urban persistence in these areas.

The last part of the paper uses the results of the previous three parts to discuss the implications of urbanization without structural transformation for long-run development.

This paper is related to a large body of work on the relationship between urbanization and growth. While the structural transformation literature cited above portrays urbanization as a by-product of economic development, the economic geography literature suggests that agglomeration promotes growth, in both developed countries (Rosenthal & Strange, 2004; Henderson, 2005) and developing countries (Overman & Venables, 2005; Henderson, 2010). Given that urbanization is a form of agglomeration, it has thus been argued that cities could promote growth in developing countries (Duranton, 2008; Venables, 2010; McKinsey, 2011).⁷ Yet evidence that urbanization drives growth is rather weak in cross-country regression frameworks (Henderson, 2003). Whether urbanization promotes growth is likely to depend on the type of the city. This paper shows that resource exports create consumption cities, with small manufacturing and tradable service sectors. If these "missing" sectors are more likely to exhibit agglomeration effects and provide incentives for capital accumulation, consumption cities could have a relatively lower positive impact on long-run growth than other cities.⁸

This paper also contributes to the Dutch disease and resource curse literatures (Corden & Neary, 1982; Matsuyama, 1992; Sachs & Warner, 2001; Michaels, 2011). Dutch disease models suggest that a country deindustrializes when its resource sector booms. The boom shifts labor and other resources away from the manufacturing sector into the non-tradable service sector, but the net effect on urbanization is ambiguous. These models do not explain why resource-rich developing countries have urbanized with almost no industrialization. I highlight a new dimension of the resource curse – the rise of "consumption cities" that may not produce the same agglomeration economies that are found in typical urban areas.

Finally, my research is related to the literature on the respective roles of geography and history in development. The locational fundamentals theory argues that natural advantages have a long-term impact on economic activity (e.g. Sokoloff & Engerman, 2000; Davis & Weinstein, 2002), while the increasing returns theory posits that there is path dependence in the loca-

⁷McKinsey (2011) writes (p.3-19): "Africa's long-term growth also will increasingly reflect interrelated social and demographic trends that are creating new engines of domestic growth. Chief among these are urbanization and the rise of the middle-class African consumer. [...] In many African countries, urbanization is boosting productivity (which rises as workers move from agricultural work into urban jobs), demand and investment."

⁸Consumption cities arise as market towns (Von Thunen, 1826; Christaller, 1933) or as places offering better consumption amenities (Glaeser, Kolko & Saiz, 2001; Rappaport, 2008). Government cities are also consumption cities because their growth is related to the government consumption of urban goods and services (e.g., civil servants) in specific urban locations (De Long & Shleifer, 1993; Ades & Glaeser, 1995; Davis & Henderson, 2003).

tion of economic activity (e.g. Rosenthal & Strange, 2004; Henderson, 2005; Bleakley & Lin, 2012). As in Nunn & Qian (2011), areas suitable for agricultural production are more developed today. However, I look at cash crops while they consider food crops, for which different mechanisms may be at play. Here the natural advantage is transitory, not known before the arrival of cocoa cultivation and not existing after exploitation. The fact that old-producing areas are still relatively more urbanized, but not necessarily more industrialized, indicates that path dependence is possible even without much structural change.

The paper is organized as follows: The next section shows there is urbanization without structural transformation in Africa. Section 3 offers a conceptual framework for thinking about the role of resource exports in urbanization. Section 4 presents the background of cocoa production and cities in Ivory Coast and Ghana and the data used in this study. Section 5 presents the empirical strategy and results. Section 6 discusses the results. Section 7 concludes.

2. PATTERNS OF URBANIZATION IN AFRICA

In this section, I document that there are different paths of urbanization, using Africa as an example. I establish that the pattern seen in Figure 2 is robust to other factors of urbanization. I also investigate whether these different paths are associated with different sectoral compositions of the urban workforce. Interestingly, Gollin, Jedwab & Vollrath (2013) obtain the same results when using a sample of 119 (African and non-African) developing countries.

2.1 Cross-sectional Robustness Checks

Table 1 presents results of cross-sectional regressions using a sample of 46 African countries for the year 2000. The first three columns use the urbanization rate as the dependent variable, regressed on the average contribution of resource exports to GDP in 1960-2000. Columns (1) shows there is a strong positive (unconditional) effect of resource exports on urbanization. There are several alternative theories for urbanization within Africa, that may make this result spurious. A few studies argue that Africa has urbanized without it being fully explained by growth (Fay & Opal, 2000). This excessive urbanization is attributed to pull and push factors feeding rural exodus. Some argue that urban growth could be driven by rural poverty (Barrios, Bertinelli & Strobl, 2006; Poelhekke, 2010). Others have focused on theories of urban bias, arguing that urban-biased policies have led to overurbanization and primacy in poor countries (Lipton, 1977; Bates, 1981; Ades & Glaeser, 1995; Davis & Henderson, 2003). Furthermore, if resource exporters systematically use different methods for calculating urbanization rates, the results may simply reflect measurement errors (see Potts 2012 for Nigeria).

In column (2) I incorporate a number of other controls to account for several of these alternative theories. These are country area in sq km, population in thousands, rural density, population growth from 1960–2000 (in percent), a dummy if the country is landlocked, a dummy if the country is often affected by droughts, a dummy for whether the country has experienced an interstate or civil conflict since independence, and a dummy if a country has been mostly autocratic since independence. In addition, we include dummies for four different types of urban definitions used by African countries.⁹ As can be seen in the table, the inclusion of these controls does not alter the positive association of resource exports with urbanization rates. The magnitude is high: A one standard deviation increase in natural resource exports

⁹See the data appendix for a full description of the sources for all of these variables.

leads to a .62 standard deviation increase in the urbanization rate. In column (3), we control for initial conditions, i.e. the urbanization rate and the contribution of resource exports to GDP in 1960. The regression is then similar to a long-difference model: The identification comes from variation across countries over time. Results are almost unchanged.¹⁰

2.2 Composition of Urban Workforce

The evidence of the prior section established that natural resources are strongly associated with urbanization in Africa. Here I establish the second important fact regarding resources and urbanization, related to the composition of labor in urban areas. I use IPUMS census data, various census reports, labor force surveys and household surveys to recreate the sectoral composition of the urban sector for 27 African countries for the year 2000.¹¹ Although the different sources often use a standard classification of nine sectors as in McMillan & Rodrik (2011), I only focus on four sectors: *Manuf. & Fire* ("Manufacturing" and "Finance, Insurance, Real Estate and Business Services", a proxy for tradable services), *Trade & Transp.* ("Whole-sale and Retail Trade, Hotels and Restaurants" and "Transport, Storage and Communications"), *Govt Services* ("Public Administration", "Education" and "Health") and *Other Sectors* ("Agriculture, Fishing and Forestry", "Mining", "Public Utilities", "Construction" and "Community, Social and Private Services"). I run the same regression as before, except the dependent variable is the employment share of each sector, for the urban sector only (columns (4)-(7)) and the largest city only (columns (8)-(11)). I also control for the urbanization rate in 2000.

Column (4) indicates that cities in resource-rich countries have less manufacturing and tradable services, for the same urbanization rate. The magnitude is also high: A one standard deviation increase in resource exports leads to a .63 standard deviation decrease in their employment share. The employment share of trade and transportation is then larger in these cities (see column (5)). A one standard deviation increase in resource exports leads to a .47 standard deviation decrease in their employment share. The explanation is simple: Resource rich countries run large export surpluses that they use to import food and manufactured goods. I find that resource exports strongly increase the share of (non-resource) imports in GDP (not shown, but available upon request). This creates opportunities in the import and domestic trade sectors (when imported goods are "dispatched" to the rest of the country). Second, it has been documented that public employment levels are high in these countries (Lipton, 1977; Bates, 1981). I also find that the employment share of government services is larger, although this effect is not significant (see column (6)). One reason for this might be that the structural adjustment programmes in the 1980's and 1990's had a very negative effect on the provision of government services, which reduced the effect for the year 2000 (Jedwab & Darko Osei, 2012). Or there are heterogeneous effects: For example, the employment share of government services is much higher in countries that are both resource rich and

¹⁰Although the goal of this section is not to identify a causal effect of resource exports on urbanization, I also estimate a short difference model with country and region-year fixed effects. The identification comes from variations within countries over time, while simultaneously controlling for regional effects specific to some years. The short-term effect is lower than the long-term effect, 0.10** against 0.62*** (see column (3) of Table 3). This is expected if the effects are not immediate. When I also include lags of resource production, I find that the combined effects of resource exports strongly increase. Back-of-the-envelop calculations confirm that these sequential short-term effects can explain the magnitude of the long-term effect. Both methods are thus complementary.

¹¹First, I was only able to find data for 27 out of 46 African countries, but these countries account for more than 80% of the continent's population. Regressions being population-weighted, adding more countries is unlikely to change the results. Second, because the census and survey years differ from the year 2000, I always use data for the closest year. Table A.1 lists all the countries and years for which I have employment data.

not democratic. Results are robust to considering the largest city only. To conclude, resource rich countries have experienced urbanization without structural transformation.

3. THEORETICAL IMPACTS OF RESOURCE EXPORTS ON URBANIZATION

This section offers a conceptual framework for thinking about the role of resource exports in urbanization. In Model Appendix B, I adapt a new structural change model that I discuss here. As emphasized before, resource production contributes little to urban employment. For resource exports to create cities, there must be consumption or production linkages.

Assume a small economy, one production factor - labor - and four sectors: food, resources, urban tradables (tradable manufactured goods and services) and urban non-tradables (non-tradable manufactured goods and services).¹² The production of urban tradables and non-tradables takes place in the cities (the resource sector is considered as non-urban). Only urban non-tradables are not tradable. Without a green revolution, food productivity is low and the food problem constrains the reallocation of labor to the urban sectors. Now, if the country has a comparative advantage in resource exports, it can trade natural resources for food and urban tradables. If the international commodity price is higher than domestic production costs, resource exports generate a surplus for the country. With non-homothetic preferences, this surplus is disproportionately spent on urban goods and services, i.e. urban tradables and non-tradables. This creates opportunities in the urban sector, which attracts people to cities. The country urbanizes through the rise of "consumption cities".

Because greater opportunities in the urban sector attract people to cities, the theory is in line with the labor pull hypothesis, except the resource sector is the main driver of this transformation. As wealth is created in the non-urban sector and spent in the urban sector, these consumption cities arise due to demand factors. If domestic and foreign foodstuffs are imperfect substitutes and/or internal trade costs are high, making imports too expensive for the hinterland, the food sector will not entirely disappear. But if urban tradables are mostly imported from abroad, the mix of urban workers is heavily skewed towards urban non-tradables. The country urbanizes without structural transformation. One limitation of the model is that it does not permit capital accumulation in the urban tradables sector, which could increase productivity over time and change the country's comparative advantage. However, if there are dynamic externalities, the original structural of comparative advantages is reinforced over time, and it is difficult for the country to compete with countries that industrialized earlier. On the contrary, countries that specialize in the export of (urban-based) manufactured goods and tradable services urbanize with structural transformation. This specialization effect gives rise to "production cities", with a mix of workers in tradable and non-tradable sectors.

In the model, any rise in the resource rent increases the labor share of urban non-tradables. First, if the supply of natural resources is very inelastic, as it is the case for minerals (e.g., oil and diamonds) and some cash crops (e.g., cocoa), the international price is much higher than domestic production costs. The size of the surplus will then determine the urbanization rate. For countries that have a comparative advantage in urban tradables, the income effect is much smaller because these markets are much more competitive. They mostly urbanize because of the specialization effect. Second, productivity in the resource sector could be so

¹²Urban tradables consist of manufactured goods and services (finance, insurance, business services, etc.) that are internationally traded. This assumes that domestic and foreign versions of these goods and services are relatively substitutable. The urban non-tradables sector addresses the specific demand for domestic manufactured goods and services, that are not internationally traded, whether its is due to local preferences or trade costs.

high that the resource rent is like a "manna from heaven", for example when we compare countries exporting minerals and countries exporting cash crops. We expect the urbanization rate to be higher for the former. Third, any rent produces urbanization, for example aid and remittances. However, the total values of aid and remittances are 9.8 and 9.0 times smaller than the total value of resource exports for developing countries in 2010 (World Bank, 2012). Fourth, because resource rich countries are more integrated with the rest of the world, they have larger trade and transportation sectors. Fifth, government taxation augments the effect of resource exports on urbanization if the government's Engel curve for urban non-tradables (e.g., civil servants) is steeper than the private Engel curve. If the government also concentrates its urban consumption in its own region, this increases the primacy rate.

4. BACKGROUND AND DATA IN IVORY COAST AND GHANA

In this section, I discuss some essential features of the Ghanaian and Ivorian economies, and introduce the data I have collected to analyze how cash crop production has contributed to urbanization. The Data Appendix contains more details on how I construct the data.

4.1 New Data on Ivory Coast and Ghana, 1891-2000

To evaluate the impact of cash crop production on urbanization, I construct a new data set on 79 Ghanaian districts from 1891-2000 and 50 Ivorian districts from 1948-1998.¹³

I first collect data on cocoa production in tons for Ghanaian and Ivorian districts for available years. I use linear interpolation for missing years. Knowing the producer price for each year, I calculate the district value of production in year 2000 U.S. dollars. Since Ivory Coast is a major coffee exporter and since Ivorian cocoa farmers also produce coffee, I proceed similarly for this crop. In the end, I obtain an annual panel data set with the value of cash crop production (cocoa and coffee) from 1891 to 2000 in Ghana and from 1948 to 1998 in Ivory Coast.

I construct a GIS database of cities using census reports. My analysis is limited to those years for which I have urban data, which is approximatively decadal in both countries.¹⁴ Historical studies on urbanization define a city as any locality with more than 5,000 inhabitants (e.g. Bairoch, 1988; Acemoglu, Johnson & Robinson, 2002).¹⁵ Using the same approach, Ghana had 9 cities in 1901 and 324 cities in 2000, while Ivory Coast had 0 cities in 1901 and 376 cities in 1998.¹⁶ Since the production data is at the district level, I use GIS to construct district urban population for the above-mentioned years.¹⁷ I also add district total population to the Ivorian panel. This is not possible for Ghana, as the cocoa districts for which I have production data differ from the administrative districts for which population data is available.

¹³The number of Ghanaian districts has been decreasing over time, while the number of Ivorian districts has been increasing. I use various sources and GIS to reconstruct the data set using the same boundaries for the whole period; 1960 cocoa district boundaries for Ghana and 1998 administrative district boundaries for Ivory Coast.

¹⁴These years are 1891, 1901, 1911, 1921, 1931, 1948, 1960, 1970, 1984 and 2000 in Ghana, and 1901, 1911, 1921, 1931, 1948, 1955, 1965, 1975, 1988 and 1998 in Ivory Coast.

¹⁵Bairoch (1988, p.217-219) discusses why the 5,000 population threshold is optimal: "In the instances where use is made of the criterion of size (that is, in most censuses), the limit of 5,000 most nearly approximates the average of the various limits employed." For 61 developing countries using a size criterion in 2000, I find that the average threshold is 4,390 inhabitants (United Nations, 2011). 7 countries adopt a size criterion above 5,000.

¹⁶Using Google Earth, I verify that each city can be clearly identified as such for the most recent census year. I also aggregate localities that actually belong to a same agglomeration.

¹⁷Cash crop production only booms from the 1960s in Ivory Coast. I use urban data before 1948 to understand urban patterns before the boom.

I complement the data set with various statistics at the country, regional and district levels, such as economic geography (e.g., transportation networks and Euclidean distance to the coast), physical geography (e.g., soil suitability for cocoa cultivation and rainfall), employment composition and migration. This data was obtained from GIS data sets, household surveys, census microdata and reports, country-level databases and agronomic studies.

4.2 Agronomic Background on Cocoa

Cocoa is produced by consuming tropical forests (Ruf, 1995*a*,*b*; Petithughenin, 1995). Cocoa smallholders go to a patch of virgin forest and replace forest trees with cocoa trees.¹⁸ Pod production starts 5 years after planting and peaks after 20 years. Pods give beans that are dried and sent unprocessed to a port for export.

The forest initially provides agronomic benefits in growing cocoa (Ruf, 1995*b*, p.7): "weed control, soil fertility, protection against erosion, moisture retention for soil and plants, protection against disease and pests, protection against drying winds, [...] stabilizing effect on precipitation."¹⁹ After 20 years, cocoa trees become too old and need to be replaced or the farmer will face a continuous decline in yields. However, the agronomic benefits of the forest are also extinguished after 20 years of local deforestation, and replanted cocoa trees die or are much less productive (Petithughenin, 1995, p.96-97). Establishing a new cocoa farm in a cleared primary forest requires 86 days of manual labor, against 168 days for manual replanting (Ruf, 1995*b*, p.9). Also, it is twice as expensive to opt for a replanting strategy as intermediate inputs (fertilizers, pesticides) are needed to compensate for tree mortality and low yields (Ruf, 1995*a*, p.240). As a result, farmers look for a decline in pod production over the course of a few years - which normally occurs 25 years after planting - as a sign to move to a new patch of virgin forest and initiate a new production cycle. These agronomic patterns at the plantation level, and the abundance of forested land for most of the 20th century, explain why African cocoa farmers have always preferred an extensive production strategy.

The aggregation of household migration patterns gives rise to regional cocoa booms that last several decades (Ruf, 1995*a*, p.190-203). Regional production at a given point in time is equal to cocoa land area times cocoa yields. Cocoa land area depends on the forest endowment and the number of farmers, while yields reflect soil nutrients and the age distribution of trees. Booming regions experience massive in-migration of farmers. In the words of Ruf (1995*b*, p.15), "It seems that all it takes is for people to see money from the first sale of a crop in the hands of the first migrant planters before a cocoa migration and boom is triggered". This is for several reasons. First, land is cheap to buy.²⁰ Second, farmers do not need much capital to start their own plantation (Ruf, 1995*b*, p.22); they only use land, axes, machetes, hoes, cocoa beans and labor. Third, farmers can easily intercrop cocoa with food crops such as cassava, taro and yam. Lastly, yields are high in the early years of production.

A few decades later, trees are old, yields have decreased and regional production declines, although production decreases can be slowed or even reversed if formerly protected forests in the region are opened to cultivation. Cocoa cultivation becomes less profitable. Ruf (1991, p.87) writes: "Planting cocoa gives social status, reflecting the ownership of capital yielding

¹⁸The involvement of multinationals in cash crop production in Africa is very recent.

¹⁹Ruf (1995*b*, p.95-100) describes how rainfall has diminished and become more erratic in old producing areas. ²⁰Ruf (1995*a*, p.252-260) documents how the land price is initially low in unexploited forests. Migrant farmers buy large amounts of land from the chiefs of forest tribes, which causes the price to rise. This land colonization

process was encouraged by the government. Ivorian President Houphouët-Boigny liked to say that "land belongs to him who cultivates it."

a huge profit. It's the golden age [...]. Then comes the phase of ageing cocoa trees: owning an old farm, attacked by insects, bearing a less valuable product, does not give any status anymore. Everything happens as if a biological curse [...] was inherent to any golden age, as if a recession should succeed any cocoa boom." Young adult males of producing households move to a new forest, participating in a new cycle elsewhere, while the old members remain on the old farm which they convert into farmland for food production.²¹

When the forest is exhausted, cocoa moves to another country or continent. Production was dominated by Latin American countries till the early 20th century, then moved to Africa and is now spreading in Asia (Ruf, 1995*a*, p.63-70). Economic and political factors can accelerate or decelerate regional cycles (Ruf, 1995*a*, p.300-359): changes in the international and producer prices, land regulations, migration policy, demographic growth, etc.

4.3 The Cash Crop Revolution in Ivory Coast and Ghana

Ghana and Ivory Coast have been two leaders of the African "cash crop revolution". They are the largest cocoa producers, and cocoa has been the motor of their development (Teal, 2002; Cogneau & Mesplé-Somps, 2002). Production boomed after the 1920s in Ghana and the 1960s in Ivory Coast. The cocoa boom was accompanied by a coffee boom in Ivory Coast, as cocoa farmers also produce coffee there.²² Cocoa and coffee have accounted for 60.2% of exports and 20.6% of GDP in Ivory Coast in 1948-2000, while cocoa has amounted to 56.9% of exports and 12.1% of GDP in Ghana over this period.

The South of each country was covered with tropical forest, while the North is savanna. In the Southern forest, some areas were more suitable for cocoa and coffee cultivation due to richer soil nutrients. Figure 3 shows *highly suitable* and *poorly suitable* districts for cocoa and coffee.²³ Figure 4 displays the total value of cash crop production in 1891-2000 for each Ghanaian district and in 1948-1998 for each Ivorian district. The comparison of Figures 3 and 4 confirms that cash crop production has been concentrated in highly suitable districts.

Cocoa was introduced to Ghana by missionaries in 1859, but production did not develop before 1900 (Hill, 1963). Production first spread out in the South-East of Ghana, in the vicinity of Aburi Botanical Gardens which were opened in 1890. British Governor W.B. Griffith wrote in 1888 (Hill, 1963, p.174): "It was mainly with the view of teaching the natives to cultivate economic plants in a systematic manner for purposes of export that I have contemplated for some time the establishment of an agricultural and botanical farm and garden where valuable plants could be raised and distributed in large numbers to the people." Cocoa seedlings were imported from São Tomé and distributed to local farmers. Since cocoa cultivation was very profitable, many farmers adopted the crop and production boomed. It peaked in the Eastern province in the 1930s (see Figure 3 and 5), before plummeting due to the Cocoa Swollen

²¹Ivorian cocoa farmers also produce Robusta coffee. Contrary to Arabica coffee which is grown at higher altitudes, Robusta coffee shares the same agronomic features as cocoa. It is grown in tropical forests and necessitates high levels of rainfall and humidity. Cherry production starts after 5 years and declines after 20 years, as the coffee tree becomes too old. This explains why cocoa and coffee have moved across regions together in Ivory Coast.

 $^{^{22}}$ Yet coffee production has never exceeded 300,000 tons per year. The cocoa boom was therefore three times more important than the coffee boom.

²³The cocoa soil type is obtained in GIS from digitized historical soil classification maps. I also verify that these maps are consistent with contemporary maps of the forest. The soil type is not affected by initial deforestation, but the nutrient content of the soil decreases as the forest is exploited. A district is *suitable* if more than 25% of its area consists of cocoa soils, i.e. the tropical forest. A district is *highly suitable* if more than 50% of district area consists of forest ochrosols, the best cocoa soils. A district is *poorly suitable* if it is suitable but not highly suitable.

Shoot Disease and World War II which reduced international demand. A second cycle started in the Ashanti province (see Figure 6 and 7). But low producer prices after 1958, restrictive migratory policies after 1969, and droughts in the early 1980s precipitated the end of this cycle (see Figure 8).²⁴ High producer prices from 1983 pushed farmers to launch a third cycle in the Western province, the last forest of Ghana.

It was not till the 1910s that the French authorities promoted cocoa and coffee in Ivory Coast (Ruf, 1995*a*,*b*). Ivorians were originally reluctant to grow these crops, except in "Indénié" in the East (Centre-East, see Fig. 3) where farmers heard of the wealth of Ghanaian cocoa farmers (Ruf, 1995*b*, p.29). Production did not boom until the 1960s.²⁵ Cocoa (and coffee) again moved from the East to the West (see Figures 5-8). The production of both crops is now concentrated in the South-West region, the last tropical forest of Ivory Coast.

To conclude, in both countries, cocoa production was confined to the South and started in the South-East. Due to the 25-year agronomic patterns at the plantation level, it moved westward (as it could not do otherwise).²⁶ As population growth was high and cocoa was profitable, many people specialized in it and participated in the expansion of productive land. Both countries have extracted the same quantity of cocoa in total: 24 million tons in Ghana versus 22 in Ivory Coast, that did so in a much shorter time period. As the forest is about to disappear, so will cocoa production, unless farmers switch to intensive production strategies.²⁷

4.4 The Urban Revolution in Ivory Coast and Ghana

While neither country was at all urbanized at the turn of the 20th century, their respective urbanization rates (using the 5,000 threshold) were 43.8% and 55.2% in 2000, making them two of the most urbanized African countries. Ghana started its urban transition earlier than Ivory Coast, but both experienced rapid urbanization after 1948. This is all the more impressive considering that the populations of Ghana and Ivory Coast have increased by 9.7 and 15.7 times respectively between 1900 and 2000. Ghana had 324 cities in 2000, while Ivory Coast had 376 in 1998, when 53.4% of urban inhabitants in Ghana and 54.8% Ivory Coast lived in small cities in the population range 5,000-20,000.

Defining as national cities the capital city and the second most important city, I calculate that they explain 45.7% of urban growth in 1901-2000 Ghana and 46.1% in 1948-1998 Ivory Coast.²⁸ Of the remaining urban growth, 66.3% in Ghana and 80.0% in Ivory Coast was in areas suitable for cocoa. This strong correlation between historical cash crop production and

²⁴Ghana (from 1948) and Ivory Coast (from 1960) have fixed the producer price to protect farmers against fluctuating international prices. The *Ghana Cocoa Marketing Board* (COCOBOD) and the Ivorian *Caisse de stabilisation et de soutien des prix des productions agricoles* (CSSPPA) were responsible for organizing the cocoa system. Since the producer price was below the international price, this served as a taxation mechanism of the sector.

²⁵Three factors explain this Ivorian "lateness". First, cocoa did not reach the Ghanaian border before 1920. Second, the French forced the Ivorians to grow cocoa and coffee through a system of mandatory labour (the *corvée*) and Ivorians only saw them as European crops. Third, production increased in the 1920s but the boom was stopped by the Great Depression and World War II.

²⁶Data on regional yields for post-1948 years also displays a westward movement in yields. The largest producing region always has the highest yields. For Ghana, this was Ashanti in the 1960s, Brong-Ahafo in the 1970s and Western in the 1990s. For Ivory Coast, this was Centre in the 1960s, Centre-West in the 1970s and early 1980s, and South-West in the late 1980s and 1990s. Data on regional production per rural capita confirms this analysis.

²⁷The forested surface of Ivory Coast has decreased from 15 million hectares in 1900 to 2.5 million in 2000, while it has decreased from 9 million in 1900 to 1.6 million in 2001 in Ghana.

²⁸The capital cities are Accra in Ghana, and Abidjan and Yamoussoukro in Ivory Coast, since Houphouët-Boigny made his village of birth Yamoussoukro the new capital in 1983. The second most important cities are Kumasi in Ghana and Bouaké in Ivory Coast. The growth of these cities is disconnected from the local context.

the emergence of cities is documented in Figure 4. This correlation is also spatio-temporal, as cities have followed the cash crop front (see Figures 5-8). As production moved westward, new cities appeared in the West, but cities in the East did not collapse. Our analysis must thus account for both city formation and city persistence.

5. EMPIRICAL STRATEGIES

Let $y_{d,T}$ be the urbanization rate of district d at the end of the period, in year T. The urbanization rate being a stock variable, given the fixed costs of building cities, it is a function of the total district value of resource production between the beginning (t_0) and the end (T)of the period NRX_{d,t_0-T} . If there is urban persistence, $y_{d,T}$ is also a function of the initial urbanization rate y_{d,t_0} . Controlling for the initial urban conditions allows me to study the effect of resource production on the long difference of the urbanization rate:

$$y_{d,T} = \alpha + \beta NRX_{d,t_0-T} + \theta y_{d,t_0} + u_d \tag{1}$$

I argue that the causality is unlikely to run from cities to cash crop production. First, there are few cities in areas that have not boomed yet (see Fig. 5-8). Settlement is limited in tropical forests due to tall trees, thick vegetation, high humidity and disease incidence. Farmers overcome these constraints when they achieve an income high enough to pay the fixed costs of deforestation, which is the case with cocoa. Second, cocoa cultivation does not depend on cities for the provision of capital and inputs, as it only requires forested land, axes, machetes, hoes, cocoa beans and labor, and farmers use small amounts of fertilizers and insecticides. This traditional mode of production is not conducive to a role for cities in the diffusion of technological innovations. Third, the ability of farmers to find labor does not depend on urban proximity. West African labor markets are highly integrated, with many laborers originating from northern regions or other countries. Later, I verify there are no effects of urbanization on future cocoa production. I will also show that the effects of cocoa production on urbanization are smaller in the short term, which is in line with the following sequence for settlement patterns: (1) tropical forest, (2) booming production associated with rural growth, (3) increasing urbanization over time.

Second, there could be some unobserved factors that would drive both resource production and urbanization over time. To deal with factors that could affect a district's urban growth path, I first control for a vector of district covariates. These controls are: (i) political economy: dummies for containing a national city, i.e. the capital and second largest cities, or a regional capital;²⁹ (ii) economic geography: dummies for having an international port, and for being connected to the railway network or the paved road network, all measured at independence (1958 in Ghana, 1960 in Ivory Coast), and Euclidean distance to the coast and Euclidean distance to the largest city, as these factors could drive both production and urban growth; (iii) physical geography and demography: district area, a dummy for being a coastal district, 1900-1945 average annual precipitation, mean altitude, and initial rural population (for 1931 in Ghana, 1948 in Ivory Coast), to control for pre-existing settlement patterns.³⁰

²⁹National cities (Abidjan, Bouaké and Yamoussoukro in Ivory Coast, Accra and Kumasi in Ghana) may grow for reasons unrelated to local production. Regional capitals at independence were also regional capitals during colonization. I thus capture any effect that would stem from the colonial urban organization.

³⁰The first year for which I have district rural population for the whole territory is 1931. I have data for Southern districts only in 1901.

However, there could be some unobserved factors that could still drive both resource production and urbanization. To address this concern, I instrument the total district value of cash crop production with a dummy whose value is one if the district is highly suitable for cocoa cultivation. The identification assumption is that conditionally on district geographical, political and economic characteristics, high land suitability does not affect the outcome independently of its impact on cash crop production. I also verify that the results hold when I restrict the analysis to the suitable districts only, i.e. all the districts of the tropical forest that are located in the south. The instrument then captures the effect of being relatively "more" suitable than other suitable districts. As argued before, highly suitable and poorly suitable areas are not different to the naked eye, differences in yields come from different nutrients in the soil that are not easily observable, even be experienced cocoa farmers. We thus predict a much higher resource rent in the highly suitable areas.

Another strategy is to exploit the fact that cocoa cultivation had to shift spatially over time. As explained above, cocoa farmers move to a new location every 25 years. This gives rise to district production cycles that can last up several decades. Besides, in a context of demographic growth, we observe an acceleration of the westward wave over time. When district production booms, we should expect to see a rise in urbanization. By using the panel dimension of the data and including district fixed effects, we can capture the effects of these local resource booms on local urbanization. I can also include district time trends and region-year fixed effects. The identification then comes from non-linear variations within districts over time, while simultaneously controlling for regional effects specific to some years. To see any effect for a district, it must be that cocoa production booms, across periods and relatively to the neighboring districts of the same region, in a difference-in-difference spirit. For Ivory Coast, I have panel data on district urbanization rates in 1948-1998. For Ghana, I do not have panel data on district urbanization rates in 1901-2000 because administrative boundaries have been considerably modified across years, this impeding any consistent reaggregation.

Lastly, there could be a few econometric concerns that both strategies cannot solve. For example, one could argue that logging enables cash crop production and urbanization. However, tree varieties are not different between the highly and poorly suitable areas, since productivity differences come from the soils. Restricting the sample to the suitable districts only should allow me to distinguish the effects of cash crop production and logging. Also, the export of forest products has only amounted to 10.1% and 16.1% of exports in Ghana and Ivory Coast in 1948-2000. Logging has been dominated by a few parastatal companies, and profits have been repatriated to the capital rather than spent locally. So we should expect no "local" effect of logging on urbanization. Lastly, even if there were such local channels, this would not alter the message of the paper that resource exports drive urbanization, as forestry exports belong to this category. My coefficients then capture the effects of cocoa, coffee and forestry.

5.1 Results

Table 2 presents the main results for 1948-1998 Ivory Coast (columns (1)-(4)) and 1901-2000 Ghana (columns (5)-(8)). In columns (1) and (5), I show the unconditional results, for which there is a strong positive effect of district resource production on district urban growth. The results are not modified when I include the geographical, political and economic controls, as shown by columns (2) and (6). In columns (3) and (7), I instrument the value of cash crop production by the highly suitable district dummy. In the first stage, I find a strong effect of the dummy on resource production, but no effect of initial urban conditions. This confirms that the early colonial cities did not play a role in the diffusion of cocoa cultivation. The

effect of cash crop production on urbanization increases to 11.7-12.0 percentage points. The Kleibergen-Paap rk Wald F Stats are high, above 15. The IV estimates indicate that the effects of resource production were downward biased before. The magnitude of these effects is high: I find that one standard deviation in cash crop production increases the urbanization rate by a .54-.57 standard deviation. In columns (4) and (8), I restrict the sample to the suitable districts only, i.e. the districts of the tropical forest. The instrument captures the effect of being relatively "more" suitable than other suitable districts. Although the point estimates decrease, they are not significantly different from the standard IV estimates.

I run a few robustness checks, whose results are presented in Table 2. Column (1) reproduces the main results from Table 1 (see column (3)). In column (2), I include the interacted effect of the high suitability dummy and the initial urbanization rate. I thus verify that there is no effect of initial urban conditions on resource production for highly suitable areas, in the first stage. In column (3), I use the share of district area that is highly suitable as the instrument instead of a dummy for being above the 50% cut-off. In column (4), I include 10 regional fixed effects for each country. In column (5), I account for spatial autocorrelation by reporting Conley standard errors using a distance threshold of 100 km.³¹ In column (6), I include the total regional production, which indicates that resource production only has local linkages. In column (7), I drop the national cities, i.e. the capital and the largest cities. In column (8), I use another urban outcome, which is the number of cities in year *T*, controlling for the number of cities in year *t*₀. Cash crop production has a strong positive effect on the number of cities, as already observed in Figures 4-8.

I show the results for the fixed effects approach in Table 3. There are 50 districts and 5 years for Ivory Coast, hence 250 observations. As discussed above, the coefficients are likely to be smaller for the panel estimation. First, the long difference estimation relates urban change to cash crop production over a very long period, thus capturing the long-term effect of the latter on the former, while the panel estimation captures the short-term effect. Second, due to the inclusion of district fixed effects, the short-term relationship between resource production and urban growth is attenuated by what is happening in the following periods, especially if cities persist although cocoa production eventually declines. Column (1) shows the unconditional results. In column (2), I include the baseline controls interacted with a linear trend. In column (3), I add the district fixed effects. The point estimate, 3.87, is exactly three times lower than the point estimate for the long difference model (11.69, see column (3) of Table 2), as expected. This result is relatively robust to including region-year fixed effects (see column (4)) and district trends (see column (5)). In column (5), the identification comes from non-linear variations within districts over time, while simultaneously controlling for regional effects specific to some years. It must be that districts urbanize as cocoa production booms, as already observed in Figures 4-8. In column (6), I also include the lag of the urbanization rate, to account for the fact that there could some urban path dependence. I also verify that the lag of the urbanization rate has no effect on the district value of cash crop production when it is used as a dependent variable instead. There is no effect of cities on cash crop production in the short-term, which reduces the concern of reverse causality. In column (7), I include the lag of the district value of cash crop production, i.e. resource production between the years t-2 and t-1. If the short-term effects are lower than the long-term effects, we should expect the short-term effect to increase over time, which is what we observe here. The coefficient is

³¹The plug-in HAC covariance matrix approach is to plug-in a covariance matrix estimator that is consistent under heteroskedasticity and autocorrelation of unknown form (Conley, 1999). By clustering standard errors at the regional level before, I was already accounting for spatial autocorrelation using a different approach.

relatively higher for the lag of resource production (4.20). The point estimate also increases for resource production (3.32), as we are now controlling for some of the long-term effects. When I add the two effects (7.52), I find that one standard deviation in cash crop production leads to a .57 standard deviation of the urbanization rate, after two periods (against .57 when using the long-difference estimation). Lastly, I investigate the short-term effects of resource production on the number of cities (see column (8)), and also find that these effects are lower than the long-term effects and that they increase over time. To conclude, the cocoa-producing districts urbanize, as the share of the urban population increases and more cities are created over time, which indicates some urban decentralization within these districts.

Lastly, I perform additional robustness checks whose results are not reported but available upon request. Results are robust to: (i) using production in volume (tons), (ii) using cocoa production only for Ivory Coast, (iii) using urban density instead of the urbanization rate, (iv) including the squares of controls, and (v) interacting controls with a non-linear trend.

5.2 Evidence on Consumption Cities [TO BE COMPLETED]

I now discuss the type of cities produced by natural resources and their long-term effects. Due to the scarcity of data in both countries, I do not have repeated measurements of income and other relevant dimensions at the district level and cannot carry out the same type of estimation as for the main results. Instead, I rely on historical data at a more aggregate spatial level and rough cross-sectional correlations on contemporary data. The sources I use are household survey and census data (the 1985-88 LSMS, 1998 and 2002 ENV in Ivory Coast; the 1987-88, 1997-98 and 2005-06 GLSS, the 2000 *Population Census* and *Facility Census* in Ghana), administrative reports, cross-country databases and agronomic studies.

5.2.1 Who Lives in These Cities?[TO BE COMPLETED]

Result 3 of the model suggests that natural resources create consumption cities. What does this imply in terms of urban employment? I compare the sectoral composition of cities in Ghana and Ivory Coast with that of cities in non-African countries with the same urbanization rate of 45-50% (using a similar urban threshold of 5,000). I use U.S. cities in 1920 and Chinese cities in 2008.³² Interestingly, Ivorian and Ghanaian cities have a relatively larger agricultural sector (17-22% against 1-3% for the U.S. and China) and a more limited industrial sector (11-24% against 43-44% in the U.S. and China). When investigating the composition of each sector, I find that they have a much smaller manufacturing sector: 7% for Ivory Coast and 16% for Ghana, against 35% in the U.S. and 28% in China. Their manufacturing exports being small, manufacturing production must address the demand for local manufactured goods. 75% of Ghanaian and Ivorian manufacturing employment consists of textile and clothing, food products, beverages and tobacco, and furniture. The same sectors account for only 30% of manufacturing employment in the U.S. and 25% in China. They also have a smaller financial and business services sector, which we could interpret as tradable services: 3% in Ivory Coast, 2% in Ghana, 6% in the U.S. and 11% in China. Given

³²I obtain historical U.S. census data from IPUMS USA. Using the 5,000 urban threshold, I estimate that the U.S. urbanization rate was 47% in 1920. The National Bureau of Statistics of China reports the urban sectoral composition on its website. It defines as a city any independent settlement of more than 3,000 inhabitants, which is quite comparable and gives an urbanization rate of 45%. I verify that my results can be generalized by extending the analysis to countries for which the urbanization is around 40-50%, the urban definition is close to the 5,000 threshold and census microdata is available. The list of these countries is available in the Data Appendix.

that Ghana and Ivory Coast do not export such services, this sector only supplies the local demand. I more generally argue that these cities are consumption cities because they arise as a result of increased demand (and not from productivity gains in the urban sector). I now investigate what gives rise to this composition, distinguishing production and consumption linkages arising from cocoa production. I abstract from including coffee to my analysis, as it is not different from cocoa.

5.2.2 Production Linkages[TO BE COMPLETED]

Backward production linkages were unlikely in this context, given the small level of urban inputs used in cocoa production. First, cocoa yields have doubled between 1960 and 2009, yet this increase has been permitted by the diffusion of high-yielding trees from the 1960s (Ruf 1995*a*, p.75-79). Aside from this innovation, production has remained traditional. Cocoa cultivation only requires forested land, axes, machetes, hoes, cocoa beans and labor. Farmers can use fertilizers and pesticides to increase yields. Yet only 6.9% of Ivorian cocoa farmers were using fertilizers in 1985-88; this share was 0.4% in Ghana in 1987-88. Meanwhile, 23.5% of Ivorian farmers were using insecticides in 1985-88 (11.6% in Ghana in 1987-88). From Ghanaian data reported by Teal, Zeitlin & Maamah (2006), I find that cocoa farmers used 3.6 kg of fertilizer and 0.14 liter of insecticide per hectare in 2002. By comparison, FAO data shows that Indonesian cocoa farmers used 95 kg of fertilizer and 0.80 liter of insecticide per hectare. The world used 94 kg of fertilizer and 3 liters of insecticide per hectare considering all crops. The data also reveals that both countries import their consumption of chemical fertilizers and insecticides.

Forward production linkages were also very unlikely in this context. First, there is no local processing of the crop. Cocoa farmers harvest cocoa pods during the peak season, which are opened to collect fresh cocoa beans. These are fermented between banana leaves, and dried by being spread in the sun on mats. The cocoa beans are later bagged and transported to the international port for export. The whole process provides no incentive for capital investments. Ruf (1995a, p.296) writes: "Unlike rubber or palm oil, no factory is needed to export cocoa beans. This relative absence of capital and technology contributes to slow down the development of agro-industries." Chocolate manufacturing is highly capital-intensive and requires refrigerated factories and ships for transport in high temperature environments. These constraints and the failure of African countries to boost manufacturing due to rent-seeking could explain why cocoa processing did not develop. I find that Ghana and Ivory Coast were responsible for 49.9% of cocoa exports in 2008, but only 0.9% of chocolate exports (FAO, 2012). Second, the agronomic literature has shown how the proceeds of cocoa farming were reinvested in buying new land, building houses and sending children to school, rather than starting new sectors (Hill, 1963; Ruf, 1995a). The cocoa sector is dominated by a myriad of smallholders, who are reluctant to deposit their savings in the formal bank system (Ruf, 1995a, p.379). Profits were thus unlikely to fund industrial projects. Third, the logistics of cocoa exports involves local and regional depots and transportation companies. Since their activity is urban-based, this could translate into more urbanization. But I find that people working in this subsector only represent 0.8% of the tertiary sector in the forest cities of Ivory Coast (1985-88).

5.2.3 Consumption Linkages[TO BE COMPLETED]

I argue that cocoa farmers spend their rising income on urban goods and services. Ruf (1995a) writes (p.379): "The possibility to enter the cocoa sector without much capital contributes to limit investments and fuels consumption fever." First, the influx of cocoa-producing households usually account for two thirds of population change in booming regions.³³ Second, cocoa farmers are wealthier than the non-cocoa farmers of the same region. Using household survey data, I regress household expenditure on a dummy equal to one if the household produces cocoa and I include village fixed effects to compare cocoa and non-cocoa farmers within the same locality. Cocoa (and coffee) farmers are 33.4% wealthier than other farmers in the forested areas of Ivory Coast in 1985-88. In the forest regions of Ghana, they are 22.1% wealthier in 1987-88.³⁴ Third, I find that cocoa farmers allocate around 30% of total consumption to home food production (mostly starchy roots, vegetables and cereals), 30% to food expenses (mostly "tasty" food such as seafood, cereals, sweets and meat) and 40% to other goods and services (mostly clothing, transfers and events, education, health and housing). Although I cannot identify which good is urban per se, I assume that food and non-food expenses imply the growth of the urban sector.³⁵ To summarize, cocoa farmers account for around 2/3 of population change in a booming region, are around 30% wealthier than other farmers, and spend around 70% of their income on urbanizing goods. If production had not boomed and if cocoa farmers had been counter-factually replaced by a number 50% lower of non-cocoa farmers, back-of-the-envelope calculations suggest that the aggregate income spent on urban goods and services would have been 2.9 times lower.³⁶ How this can be related to urban employment growth is difficult to say, but the evidence generally suggests large consumption linkages.37

5.2.4 Government Taxation and Redistribution [TO BE COMPLETED]

I now investigate whether state capture affects the composition and spatial distribution of urbanization effects (Result 4). As explained before, the government has respectively captured 40.5% and 49.9% of cocoa (and coffee) exports in Ivory Coast. The cocoa tax accounted for around 20% of government spending in 1961-2000. Also, around two thirds of government spending corresponds to government consumption (public employment), while the rest consists of government investments (infrastructure). In cities of producing districts, I find that public employment represented around 10% of total employment in the 1980s, in line

³³They respectively account for 79.7% and 62.9% of population change in the Ivorian Centre-West region between 1988 and 1998 and the Ghanaian Western province between 1984 and 2000.

³⁴In Jedwab & Moradi (2011), we describe a very similar story for the Eastern province in 1931. As production boomed there, there was a massive influx of migrants who were getting rich by working on cocoa farms. Cardinall (1931) writes (p.84): "An influx of strangers drawn here as it were to El Dorado has opened up the country to an extent that no man could have foreseen as possible within so short a period."

³⁵Dercon & Hoddinott (2005) show on Ethiopian data that rural households go to the nearest market town to: (i) buy 47% of crop inputs, (ii) sell a large share of crop production, (iii) get non-agricultural income by selling artisanal products, and (iv) purchase 55% of their consumables.

³⁶Assuming that cocoa farmers would have been replaced by half less non-cocoa farmers without any cocoa boom is a more than reasonable hypothesis, given the high fixed costs of deforestation. As argued earlier, farmers were willing to overcome those costs only when agricultural production was profitable, which was the case with cocoa and coffee.

³⁷Another issue is why so many cocoa (and non-cocoa) farmers live in town. As argued earlier, living in the forest can be difficult and the prevailing production technologies may allow farmers to commute to their land only when necessary. Cities also offer better consumption amenities, such as leisure and recreational activities, durable housing and infrastructure.

with the national average. These cities thus did not grow as administrative centres. But I find that villages and cities of the old producing districts have better infrastructure. This could make these villages grow and pass the 5,000 threshold, or this could contribute to the growth of existing cities. I first create district measures of infrastructure provision for Ghana in 2000. I then create district dummies if cocoa production has boomed in the 1980s or 1990s (the newly producing districts) or before the 1980s (the old producing districts).³⁸ Finally, I regress the various measures of infrastructure on these district dummies, including the same baseline controls as before. Results from Table 5 confirm that rural and urban inhabitants of old producing districts are more likely to have access to health facilities (col. (1)-(4)), schools (col. (5)-(10)), communications (col. (11)-(14)), utilities (col. (15)-(18)) and roads (col. (19)), while newly producing districts do not seem to have relatively better infrastructure. Results for Ivory Coast are similar (not reported but available upon request). Public employment accounts for around 20% of total employment in national cities (Abidjan, Bouaké and Yamoussoukro in Ivory Coast, Accra and Kumasi in Ghana), which indicates their role as administrative centres. Besides, around two thirds of the total civil servant wage bill is concentrated in these cities. As we know the total amount of the cocoa tax, its spatial allocation rule (based on the wage bill), and assuming one million dollars leads to 74 additional urban inhabitants (see section 4.3), I find that state capture respectively explains 29.5% and 48.4% of the urban growth of national cities in Ghana and Ivory Coast. Indeed, Abidjan is the fifth largest city in Sub-Saharan Africa, despite the fact that Ivory Coast is only the thirteenth largest country.

6. DISCUSSION

The economic geography literature suggests that agglomeration promotes growth, in both developed countries (Rosenthal & Strange, 2004; Henderson, 2005) and developing countries (Overman & Venables, 2005; Henderson, 2010). Given that urbanization is a form of agglomeration, it has thus been argued that cities could promote growth in developing countries (Duranton, 2008; Venables, 2010; McKinsey, 2011). Yet evidence that urbanization drives growth is rather weak in cross-country regression frameworks (Henderson, 2003).

Whether urbanization promotes growth is likely to depend on the type of the city, and the sectors in the city, if different sectors have different effects on long-run growth. This paper shows that resource exports create consumption cities, with small manufacturing and tradable service sectors. There are several channels through which this type of urbanization could have a relatively lower effect on long-run growth. First, there could be dynamic externalities in the urban tradables sector, as in Young (1991) or Matsuyama (1992), and the resource rich countries and cities could experience lower rates of overall technical progress. Second, incentives for productivity improvement could be greater due to the threat of global competitors, as in Galdon-Sanchez & Jr. (2002), Bustos (2011) or Aw, Roberts & Xu (2011). Third, there could be large intersectoral linkages from the urban tradables sector. Fourth, the urban tradables sector could provide more incentives for human capital accumulation Galor & Mountford (2008), which would then affect overall productivity. Lastly, cities in resource rich countries could be more corrupt and associated with a high degree of income inequality. This would then have a significant effect on the quality of institutions at the national level, that

 $^{^{38}}$ I use infrastructure data from the 2000 *Facility Census* and *Population Census*. As they adopt the administrative district decomposition in 2000 (N = 110), which differs from the one I have used before, I identify new and old producing districts by studying regional patterns of production. Production boomed in the districts of the *Western* province in the 1980s or 1990s.

would affect overall productivity.

7. CONCLUSION

This paper documents several new facts regarding the processes of urbanization and structural transformation using cross and within country evidence. While most developing countries (e.g., in Asia) have urbanized with structural transformation (i.e., a shift of labor towards manufacturing and tradable services), resource rich developing countries (e.g., in Africa) have experienced urbanization without structural transformation. The Engel curve implies that resource windfalls are disproportionately spent on urban goods and services, which gives rise to consumption cities. If manufactured goods and tradable services are imported from abroad, the mix of urban workers is heavily skewed towards non-tradable services.

In order to estimate the causal effect of resource exports on the extent and type of urban growth, I use a natural experiment in Ivory Coast and Ghana. Using new district data spanning one century and two identification strategies (i.e., an instrumental variables strategy and a fixed effects approach), I find a strong causal effect of the production of cocoa, a rural-based natural resource, on the growth of cities. However, I do not find that these cities industrialized over time. Therefore, these countries urbanized without structural transformation.

This paper makes three contributions to our understanding of the relationship between natural resources, urbanization and economic development.

The first contribution is establishing that there are many countries for which the processes of urbanization and structural transformation are disconnected. Africa offers a perfect example of this type of urbanization. In contrast to standard theories of structural transformation, Africa did not urbanize following a green revolution or an industrial revolution, but as a result of natural resource exports.

Secondly, I provide evidence on the relationship between economic development and urbanization. This paper shows that resource exports leads to urbanization. Whether urbanization in turn promotes growth might depend on the type of cities created. I argue that resource exports create consumption cities, with small manufacturing and tradable service sectors. These different types of cities may have different long-run growth effects. If these "missing" sectors are more likely to exhibit agglomeration effects and provide incentives for skill accumulation, consumption cities will have a relatively small impact on long-run growth.

Finally, this paper confirms that resource exports have positive economic effects in the short term, as producing regions accumulate cities. However, I highlight a new dimension of the resource curse – the rise of consumption cities may not produce the growth effects that are found in typical urban areas.

This paper leaves several open questions that warrants further study. Future research should explore why most resource rich countries in Africa, in particular, Ghana and Ivory Coast, have been unable to develop a comparative advantage in the non-resource sectors. As we acquire more data, later studies can investigate whether these consumption cities will evolve into production cities over time, as they did in the United States and a few Southeast Asian countries.

REFERENCES

- Acemoglu, Daron, Simon Johnson, and James A. Robinson. 2002. "Reversal Of Fortune: Geography And Institutions In The Making Of The Modern World Income Distribution." *The Quarterly Journal of Economics*, 117(4): 1231–1294.
- Ades, Alberto F, and Edward L Glaeser. 1995. "Trade and Circuses: Explaining Urban Giants." *The Quarterly Journal of Economics*, 110(1): 195–227.
- Alvarez-Cuadrado, Francisco, and Markus Poschke. 2011. "Structural Change Out of Agriculture: Labor Push versus Labor Pull." *American Economic Journal: Macroeconomics*, 3(3): 127–58.
- Aw, Bee Yan, Mark J. Roberts, and Daniel Yi Xu. 2011. "R&D Investment, Exporting, and Productivity Dynamics." *American Economic Review*, 101(4): 1312–44.
- **Badiane, Ousmane.** 2011. *Agriculture and Structural Transformation in Africa*. Center on Food Security and the Environment: Stanford Symposium Series on Global Food Policy and Food Security in the 21st Century.
- **Bairoch, Paul.** 1988. *Cities and Economic Development: Frow the Dawn of History to the Present.* Chicago: The University of Chicago Press.
- Barrios, Salvador, Luisito Bertinelli, and Eric Strobl. 2006. "Climatic Change and Rural-Urban Migration: The Case of Sub-Saharan Africa." *Journal of Urban Economics*, 60(3): 357–371.
- **Bates, Robert.** 1981. *Markets and States in Tropical Africa: The Political Basis of Agricultural Policies.* Berkeley: University of California Press.
- Bleakley, Hoyt, and Jeffrey Lin. 2012. "Portage and Path Dependence." *The Quarterly Journal of Economics*, 127(2): 587–644.
- **Bustos, Paula.** 2011. "Trade Liberalization, Exports, and Technology Upgrading: Evidence on the Impact of MERCOSUR on Argentinian Firms." *American Economic Review*, 101(1): 304–40.
- Cardinall, Allan W. 1931. The Gold Coast. Government Printer.
- **Caselli, Francesco.** 2005. "Accounting for Cross-Country Income Differences." In *Handbook of Economic Growth*. Vol. 1, , ed. P. Aghion and S. Durlauf, Chapter 9, 679–741. Elsevier.
- **Caselli, Francesco, and Wilbur John Coleman II.** 2001. "The U.S. Structural Transformation and Regional Convergence: A Reinterpretation." *Journal of Political Economy*, 109(3): 584–616.
- **Christaller, Walter.** 1933. *Central Places in Southern Germany*. Translation into English by Carlisle W. Baskin in 1966, Englewood Cliffs, NJ: Prentice-Hall.
- **Cogneau, Denis, and Sandrine Mesplé-Somps.** 2002. "L'économie ivoirienne, la fin du mirage ?" DIAL Working Papers DT/2002/18.
- **Conley, Timothy.** 1999. "GMM estimation with cross sectional dependence." *Journal of Econometrics*, 92(1): 1–45.
- **Corden, W Max, and J Peter Neary.** 1982. "Booming Sector and De-Industrialisation in a Small Open Economy." *Economic Journal*, 92(368): 825–48.

- **Davis, Donald R., and David E. Weinstein.** 2002. "Bones, Bombs, and Break Points: The Geography of Economic Activity." *American Economic Review*, 92(5): 1269–1289.
- **Davis, James C., and J. Vernon Henderson.** 2003. "Evidence on the political economy of the urbanization process." *Journal of Urban Economics*, 53(1): 98–125.
- **De Long, J Bradford, and Andrei Shleifer.** 1993. "Princes and Merchants: European City Growth before the Industrial Revolution." *Journal of Law and Economics*, 36(2): 671–702.
- **Dercon, Stefan, and John Hoddinott.** 2005. "Livelihoods, Growth, and Links to Market Towns in 15 Ethiopian villages." *International Food Policy Research Institute FCND discussion papers*, 194.
- **Duranton, Gilles.** 2008. "Viewpoint: From Cities to Productivity and Growth in Developing Countries." *Canadian Journal of Economics*, 41(3): 689–736.
- Evenson, Robert E., and Douglas Gollin. 2003. "Assessing the Impact of the Green Revolution, 1960 to 2000." *Science*, 300(5620): 758–762.
- FAO. 2012. FAOSTAT. Rome: Food and Agricultural Organization.
- Fay, Marianne, and Charlotte Opal. 2000. "Urbanization without growth : a not-souncommon phenomenon." The World Bank Policy Research Working Paper Series 2412.
- Galdon-Sanchez, Jose E., and James A. Schmitz Jr. 2002. "Competitive Pressure and Labor Productivity: World Iron-Ore Markets in the 1980's." *American Economic Review*, 92(4): 1222–1235.
- Galor, Oded, and Andrew Mountford. 2008. "Trading Population for Productivity: Theory and Evidence." *Review of Economic Studies*, 75(4): 1143–1179.
- Glaeser, Edward L., Jed Kolko, and Albert Saiz. 2001. "Consumer city." *Journal of Economic Geography*, 1(1): 27–50.
- **Gollin, Douglas, Remi Jedwab, and Dietrich Vollrath.** 2013. "Urbanization with and without Structural Transformation." Unpublished manuscript, George Washington University, Department of Economics.
- **Gollin, Douglas, Stephen L. Parente, and Richard Rogerson.** 2002. "The Role of Agriculture in Development." *American Economic Review*, 92(2): 160–164.
- **Gollin, Douglas, Stephen L. Parente, and Richard Rogerson.** 2007. "The food problem and the evolution of international income levels." *Journal of Monetary Economics*, 54(4): 1230–1255.
- Hansen, Gary D., and Edward C. Prescott. 2002. "Malthus to Solow." American Economic Review, 92(4): 1205–1217.
- Harris, John R., and Michael P. Todaro. 1970. "Migration, Unemployment & Development: A Two-Sector Analysis." *American Economic Review*, 60(1): 126–42.
- Henderson, J. Vernon. 2003. "The Urbanization Process and Economic Growth: The So-What Question." *Journal of Economic Growth*, 8(1): 47–71.
- Henderson, J. Vernon. 2005. "Urbanization and Growth." In *Handbook of Economic Growth*. Vol. 1, , ed. P. Aghion and S. Durlauf, Chapter 24, 1543–1591. Elsevier.

- Henderson, J. Vernon. 2010. "Cities and Development." *Journal of Regional Science*, 50(1): 515–540.
- Hill, Polly. 1963. *The migrant cocoa-farmers of southern Ghana: a study in rural capitalism.* Cambridge: Cambridge University Press.
- Jedwab, Remi, and Robert Darko Osei. 2012. "Structural Change and Development in Ghana." Report prepared for the project "Structural Change in Developing Countries", Coorganized by Margaret McMillan and Dani Rodrik.
- Jedwab, Rémi, and Alexandre Moradi. 2011. "Transportation Infrastructure and Development in Ghana." PSE Working Papers 2011-24.
- Lewis, Arthur. 1954. "Economic Development with Unlimited Supplies of Labour." *The Manchester School*, 22(2): 139–191.
- Lipton, Michael. 1977. Why Poor People Stay Poor: Urban Bias in World Development. Cambridge: Harvard University Press.
- Lucas, Robert E. 2004. "Life Earnings and Rural-Urban Migration." *Journal of Political Economy*, 112(S1): S29–S59.
- Lucas, Robert E. 2009. "Trade and the Diffusion of the Industrial Revolution." *American Economic Journal: Macroeconomics*, 1(1): 1–25.
- Matsuyama, Kiminori. 1992. "Agricultural Productivity, Comparative Advantage, and Economic Growth." *Journal of Economic Theory*, 58(2): 317–334.
- **McKinsey.** 2011. *Lions on the Move: The Progress and Potential of African Economies.* London : Mc Kinsey Global Institute.
- McMillan, Margaret S., and Dani Rodrik. 2011. "Globalization, Structural Change and Productivity Growth." NBER Working Papers 17143.
- Michaels, Guy. 2011. "The Long Term Consequences of Resource-Based Specialisation." *The Economic Journal*, 121(551): 31–57.
- Michaels, Guy, Ferdinand Rauch, and Stephen J. Redding. 2012. "Urbanization and Structural Transformation." *The Quarterly Journal of Economics*, 127(2): 535–586.
- Nunn, Nathan, and Nancy Qian. 2011. "The Potato's Contribution to Population and Urbanization: Evidence from an Historical Experiment." *The Quaterly Journal of Economics*, 126(2).
- **Overman, Henry, and Anthony J. Venables.** 2005. "Cities in the Developing World." Centre for Economic Performance, LSE CEP Discussion Papers dp0695.
- **Petithughenin**, **Philippe**. 1995. "Regeneration of Cocoa Cropping Systems: The Ivorian and Togolese Experience." In *The Economics of Cocoa Supply*., ed. Francois Ruf and P S Siswputranto. Cambridge: Woodhead Publishing Limited.
- **Poelhekke, Steven.** 2010. "Urban Growth and Uninsured Rural Risk: Booming Towns in Bust Times." *Journal of Development Economics*, In Press.
- **Potts, Deborah.** 2012. "Challenging the Myths of Urban Dynamics in Sub-Saharan Africa: The Evidence from Nigeria." *World Development*, 40(7): 1382–1393.

- **Rappaport, Jordan.** 2008. "Consumption Amenities and City Population Density." *Regional Science and Urban Economics*, 38(6): 533–552.
- Restuccia, Diego, Dennis Tao Yang, and Xiaodong Zhu. 2008. "Agriculture and Aggregate Productivity: A Quantitative Cross-Country Analysis." *Journal of Monetary Economics*, 55(2): 234–250.
- **Rosenthal, Stuart S., and William C. Strange.** 2004. "Evidence on The Nature and Sources of Agglomeration Economies." In *Handbook of Regional and Urban Economics*. Vol. 4, , ed. J. V. Henderson and J. F. Thisse, Chapter 49, 2119–2171. Elsevier.
- **Ruf, François.** 1991. "Les crises cacaoyères. La malédiction des âges d'or ?" *Cahiers d'études africaines*, 31(121): 83–134.
- Ruf, François. 1995a. Booms et crises du cacao: Les vertiges de l'or brun. Paris: Ministère de la Coopéeration, CIRAD-SAR and Karthala.
- **Ruf, Francois.** 1995b. "From "Forest Rent" to "Tree Capital": Basic Laws of Cocoa Supply." In *The Economics of Cocoa Supply*., ed. F. Ruf and P S Siswputranto. Cambridge: Woodhead Publishing Limited.
- Sachs, Jeffrey D., and Andrew M. Warner. 2001. "The Curse of Natural Resources." *European Economic Review*, 45(4-6): 827–838.
- Schultz, Theodore W. 1953. The Economic Organization of Agriculture. New York: McGraw-Hill.
- **Sokoloff, Kenneth L., and Stanley L. Engerman.** 2000. "Institutions, Factor Endowments, and Paths of Development in the New World." *Journal of Economic Perspectives*, 14(3): 217–232.
- **Teal, Francis J.** 2002. "Export Growth and Trade Policy in Ghana in the Twentieth Century." *The World Economy*, 25(9): 1319–1337.
- Teal, Francis J., Andrew Zeitlin, and Haruna Maamah. 2006. Ghana Cocoa Farmers Survey 2004: Report to Ghana Cocoa Board. CSAE-Oxford University.
- **United Nations.** 2011. *World Urbanization Prospects: The 2011 Revision*. New York: United Nations, Department of Economic and Social Affairs.
- Venables, Anthony J. 2010. "Economic Geography and African Development." Papers in Regional Science, 89(3): 469–483.
- **Von Thunen, Johann Heinrich.** 1826. *Isolated State*. Translation into English by Carla M. Wartenberg in 1966. New York: Pergamon Press.
- World Bank. 2012. World Development Indicators. Washington D.C.: World Bank.
- Yang, Dennis, and Xiaodong Zhu. 2010. "Modernization of Agriculture and Long-Term Growth." *IZA Discussion Papers*, 5239.
- Yi, Kei-Mu, and Jing Zhang. 2011. "Structural Change in an Open Economy." *Federal Reserve Bank of Minneapolis Staff Report*, 456.
- Young, Alwyn. 1991. "Learning by Doing and the Dynamic Effects of International Trade." *The Quarterly Journal of Economics*, 106(2): 369–405.

Figure 1: Urbanization and the Contribution of Manufacturing and Services to GDP for Developing Countries, by Type of Countries, 2000.



Notes: This figure shows the relationship between the urbanization rate (%) and the contribution of manufacturing and services to GDP (%) for 119 developing countries across four areas in 2000: Asia (30), Latin America and the Caribbean (26), Sub-Saharan Africa (46), and Middle-East and North Africa (17). Countries for which natural resource exports NRX (fuels, mining, forestry and cash crops) account for less than 10% of GDP in 2000 are in black. Countries for which natural resource exports NRX account for more than 10% of GDP in 2000 are in grey. The quadratic fit is estimated using population in 2000 as weights. See Data Appendix for data sources and construction of variables.



Figure 2: Urbanization and Natural Resource Exports in Africa, 1960-2000.

Notes: This figure shows the relationship between the urbanization rate (%) in 2000 and the average contribution of natural resource exports (fuels, mining, forestry and cash crops) to GDP (%) in 1960-2000 for 46 Sub-Saharan African countries. The linear fit is estimated using population in 2000 as weights. See Data Appendix for data sources and construction of variables.



Figure 3: Cocoa Soil Suitability and Regional and District Boundaries.

Notes: This figure shows the regional and district boundaries (N = 50 in Ivory Coast, N = 79 in Ghana), and the districts suitable for cocoa cultivation. The South of each country was historically covered with dense tropical forest. A district is defined as *suitable* if more than 25% of its area consists of cocoa soils (the tropical forest). A district is defined as *highly suitable* if more than 50% of its area consists of forest ochrosols, the best cocoa soils. It is defined as *poorly suitable* if it is suitable, but not highly suitable. See Data Appendix for data sources and construction of variables.



Figure 4: Value of Cash Crop Production (1900-2000) and Cities (2000).

Notes: This figure shows the district total value (in thousands of year 2000 U.S. dollars per sq.km.) of cocoa production in Ghana from 1891 to 2000, and cocoa and coffee production in Ivory Coast from 1948 to 2000. It also shows the cities in 2000 (1998 for Ivory Coast), defined as the localities whose population is superior to 5,000 inhabitants. See Data Appendix for data sources and construction of variables.



Figure 5: District Density of Cocoa Production and Cities in 1948.

Notes: This figure shows the district density of cocoa production (in tons per sq.km.) of cocoa production in Ghana and Ivory Coast in 1948. It also shows the cities in 1948, defined as the localities whose population is superior to 5,000 inhabitants. See Data Appendix for data sources and construction of variables.



Figure 6: District Density of Cocoa Production and Cities in 1960-1965.

Notes: This figure shows the district density of cocoa production (in tons per sq.km.) of cocoa production in Ghana in 1960 and Ivory Coast in 1965. It also shows the cities for the same years, defined as the localities whose population is superior to 5,000 inhabitants. See Data Appendix for data sources and construction of variables.



Figure 7: District Density of Cocoa Production and Cities in 1970-1975.

Notes: This figure shows the district density of cocoa production (in tons per sq.km.) of cocoa production in Ghana in 1970 and Ivory Coast in 1975. It also shows the cities for the same years, defined as the localities whose population is superior to 5,000 inhabitants. See Data Appendix for data sources and construction of variables.



Figure 8: District Density of Cocoa Production and Cities in 1984-1988.

Notes: This figure shows the district density of cocoa production (in tons per sq.km.) of cocoa production in Ghana in 1984 and Ivory Coast in 1988. It also shows the cities for the same years, defined as the localities whose population is superior to 5,000 inhabitants. See Data Appendix for data sources and construction of variables.

1 LONG-DIFFEREN(TABLE 1 CE ANE	l: URB.) CROS	ANIZATI S-SECTI	ON AND ONAL E	NATUR STIMAT	LAL RES IONS, S	OURCE] UB-SAH	EXPORI ARAN A	'S FRICA,	1960-20	000
Dependent Variable:	n I	rb. Rate ((%)	Empl. S	hare (%),	Urban Sec	tor Only	Empl. S	hare (%),	Largest C	ity Only
			Long Diff. Model	Manuf. & Fire	Trade & Transp.	Govt Services	Other Sectors	Manuf. & Fire	Trade & Transp.	Govt Services	Other Sectors
	(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)	(10)	(11)
Natural Resource Exports (% of GDP, Av. 1960-2000)	0.86*** (0.06)	0.77** (0.16)	0.62^{***} (0.10)	-0.37** (0.08)	0.32* (0.12)	0.12 (0.12)	-0.07 (0.12)	-0.37** (0.09)	0.31^{*} (0.13)	0.15 (0.14)	-0.09 (0.23)
Urbanization Rate (%, 2000)				0.61^{***} (0.08)	-0.22 (0.16)	0.04 (0.14)	-0.44*** (0.06)	0.57** (0.11)	-0.43 (0.23)	-0.15 (0.09)	0.02 (0.20)
Controls Observations	N 46	Y 46	Y 46	Y 27	Y 27	Y 27	Y 27	Y 27	Y 27	Y 27	Y 27
R-squared	0.40	0.63	0.88	0.74	0.78	0.16	0.49	0.89	0.85	0.03	0.50
<i>Notes</i> : OLS regressions using cros in parentheses; * p<0.10, ** p<0 the average contribution of nature urbanization rate (%) in 2000. If 1960. The regression is a long-diff have data for 27 countries that ac Estate and Business Services', (ii) ' "Public Administration', "Education "Community, Social and Private Se population-weighted and include t area), annual population growth r "threshold plus condition"), the pol equal to one if the country is landl if the country's average combined interstate or civil conflict since ind and construction of variables.	3.8-sectional (3.05, *** p al resource (al resource (ference mod ccount for 8 "Trade & Tra "Trade & Tra "Trate in 1960 pulation thr pulation thr llocked, a du polity score lependence. 	lata for 46 (2011. Thei exports (fue 3), we cont lel. The der lel. The	Sub-Saharan e are four reg els, mining, foi rol for initial bendent variab e continent's F olesale and Re v) Other Secto v) Other Secto t the country l four dummies n this type of t to one if the bendence is lov 3)-(11) also in	African countripolations: Wester restry and ca restry and ca conditions, i, ole in column, oppulation. T and the in column (8)-(1) alumns (8)-(1) alumns (8)-(1) alumns (8)-(1) evel: area (s for each tyr definition is u country has ever than -5 (nclude a dum	trifes in 1960 n Africa, Cer sh crops) to .e. the urban s (4)-(7) is th The four secto otels and Res otels at otels at otels at otel at otel at otels at otels at otels at otels at otel at otels at otels at otel at otel at ote	2000. Robu dDP (%) in hization rate he employme ors are: (i) M rataurants" and ration (1000s efinition ("ac ny equal to o nore drough s autocratic) one if the lary	st standard er lastern Africa 1960-2000. T (%) and the nt share of foi fanuf. & Fire: farming and ure of these se ure of these se trunistrative", ne if the coun s than the res , and a dumm gest city is the	rors clustere and Souther he depender contribution "Manufactu "Manufactu torage and (Quarrying", ctors for the y (1000s of "threshold", try is a small t of Africa ir y equal to on capital city.	d at the reg in Africa. Th it variable ir of resource tectors for th uring" and "F Communicati largest city (in largest city (in 1960-2000) n i f the couu See Data Ap	ional level al e variable of columns (1 exports to G e urban secti inance, Insu ons", (iii) Go ins", (iii) Go ins", "Constru- naly. All regr ion per sq ku nd administi 0,000 sq km) a dummy e ntry has expe pendix for d	e reported interest is)-(3) is the iDP (%) in or only. We cance, Real <i>vt Services:</i> action" and essions are active", and a dummy qual to one crienced an ata sources

LONG-DIFF	ERENCE	ESTIMAT	IONS, IVC	DRY COAST	AND GHA	NA, 1901	-2000	
Country, Period:		IVORY COA	ST, 1948-19	98		GHANA	, 1901-2000	
Dependent Variable:	מ	rbanization	Rate (%) in	1998		Irbanization	Rate (%) in	2000
Method: Restricted Sample:	OLS	OLS	N	IV Suitable Only	OLS	OLS	N	IV Suitable Only
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Panel A: Main Equation Value of Cash Crop Exports (Standard Score, Same Period)	6.62*** (0.93)	7.50*** (1.36)	11.69** (4.78)	10.68^{**} (4.70)	5.21*** (1.96)	5.54*** (1.74)	12.03** (5.29)	7.18** (3.17)
Urbanization Rate, Ex Ante (Iv. Coast = 1948, Ghana 1901))	0.62^{**} (0.25)	0.53^{**} (0.26)	0.54* (0.28)	0.46^{***} (0.011)	0.56*** (0.07)	0.12 (0.13)	0.15 (0.15)	0.76*** (0.19)
Panel B: First Stage Highly Suitable District			1_28***	1_23***			1.45**	18**
(Dummy, Ex Ante)			(0.29)	(0.23)			(0.37)	(0.29)
Urbanization Rate, Ex Ante (Iv. Coast = 1948, Ghana 1901))			-0.01 (0.02)	-0.00 (0.02)			-0.01 (0.01)	-0.03 (0.03)
Kleibergen-Paap rk Wald F Stat			19.4	27.1			15.6	23.0
Controls Observations	ΖÇ	Y Q2	Y Q	Ч 35	N QL	Ч 78	Ч 78	Ч 36
R-squared	0.46	0.60	0.57	0.63	0.20	0.57	0.53	0.61
<i>Notes:</i> OLS and IV regressions using cross-s at the regional level are reported in parentl urbanization rate (%) in 1998 for Ivory Coa and 1901 for Ghana. <i>Value of Cash Crop Ex</i> for Ghana (cocoa). <i>Highly Suitable District</i> i restrict the sample to the districts suitable fo at the district level. Political economy: dum dummies for whether the district has a pave the coast. Physical geography: district area rural population (1948 in Ivory Coast, and modurcion (cold hauvite manoanese and d	ectional data for heses; * p <0.10 st and 2000 for ' ports is the stan s a dummy equi- or cocoa cultivati mies equal to on ed road, a railro (sq km), a dist 1931 for Ghana fiamonds) in 19	r 50 districts in , ** $p<0.05$, ** Ghana. In all re- dardized distric al to one if more ion, whether the ion, whether the e if the district of ad or a port at i rict dummy for the data being 0.1-2000 See D	Ivory Coast in ** p<0.01. The gressions, we cc t total value of t total value of t total solve of d y are highly or j contains a natioi independence (being a coastal y very imperfect	1948-2000 and 78 re are ten regions 1 mitrol for initial urb cash crop productio istrict area consists poorly suitable. The nal city (the capital 1958 in Ghana, 190 district, 1900-1945 for 1901). The re- i for 1901). The re-	districts in Ghan for each country, an conditions, i.e on in 1948-2000 of forest ochrost of forest ochrost e regressions in co and second large fo in Ivory Coast 50 in Ivory Coast 50 in Ivory Coast 51 in Vory Coast 52 in Vory Coast 53 in Vory Coast 54 in Vory Coast 55 in Vory Coast 56 in Vory Coast 50 in Vory C	na in 1901-2000 The dependen for Ivory Coast ols (the best coc olumns (2)-(4) est city) or a reg est city) or a reg precipitation (i precipitation (i ana also include	0. Robust stand. t variable in col ion rate (%) in J ion rate (%) in 1 : (cocoa and coff : co soils). In col and (6)-(8) inch and (6)-(8) inch and (6)-(8) inch and (6)-(8) to stances (km) to mm), mean alti es the district to	ard errors clustered umns (1)-(4) is the 948 for Ivory Coast ce) and 1901-2000 umns (4) and (8), I ide various controls conomic geography: the largest city, and ude (m) and initial tal value of mineral

TABLE 2: URBANIZATION AND NATURAL RESOURCE EXPORTS

Dependent Variable:		Urbanizatic	n Rate (%) i	in 1998 (Ivc	ory Coast), 2	000 (Ghana		Number of
								Cities 1998/2000
Method:	N	N	2 .	N ·	2 ;	2	2	N
Kobustness Uneck:	Main	Hignlysuit. x Urb.Rate	Continuous Share	Kegional FE	Conley 100 km	Kegional Value	Urop Iop Cities	Main
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Panel A: Ivory Coast, 1948-1998								
Value of Cash Crop Exports	11.69^{**}	11.76^{**}	9.00**	13.85***	11.69^{***}	10.63^{***}	12.31^{**}	5.96***
(Standard Score, Same Period)	(4.78)	(4.92)	(4.12)	(5.10)	(4.10)	(2.66)	(4.97)	(0.88)
1st Stage: Highly Suitable District	1.28^{***}	1.29^{***}	0.02^{***}	1.15^{***}	1.28^{***}	1.33^{***}	1.25^{***}	1.28^{***}
(Indicator or Share, Ex Ante)	(0.29)	(0.28)	(0.01)	(0.27)	(0.28)	(0.29)	(0.31)	(0.29)
Panel B: Ghana, 1901-2000								
Value of Cash Crop Exports	12.03^{**}	11.62^{**}	8.57**	16.77^{**}	12.03^{**}	11.86^{*}	11.80^{**}	5.58***
(Standard Score, Same Period)	(5.29)	(2.00)	(4.34)	(8.60)	(2.86)	(6.11)	(5.78)	(1.36)
1st Stage: Highly Suitable District	1.45^{***}	1.51^{***}	0.02^{**}	1.41^{***}	1.45^{***}	1.41^{***}	1.30^{***}	1.45^{***}
(Indicator or Share, Ex Ante)	(0.37)	(0.36)	(0.01)	(0.40)	(0.40)	(0.40)	(0.36)	(0.37)
Controls	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ
Observations	50; 78	50; 78	50; 78	50; 78	50; 78	50; 78	50; 78	50; 78
Notes: OLS and IV regressions using cross- at the regional level (except in column (5)) (%) in 1998 for Ivory Coast and 2000 for urban conditions. Value of Cash Crop Expor-	sectional data) are reported i Ghana. In colu rts is the standa	for 50 districts in n parentheses; * mn (8) the depe urdized district to	n Ivory Coast in p<0.10, ** p<(endent variable i otal value of casl	1948-2000 an. 0.05, *** p<0.0 is the number of the cop product	d 78 districts in 01. The depend of cities (localit ion in 1948-200	Ghana in 1901 (ent variables in ies > 5,000 in h for Ivory Coa	 I-2000. Robust st columns (1)-(7) In all regressi st (cocoa and col 	andard errors clustered is the urbanization rate ons, I control for initial fee) and 1901-2000 for
Ghana (cocoa). <i>Highly Suitable District</i> is a interacted effect of high suitability and the	dummy equal initial urbaniz	to one if more th ation rate to see	ian 50% of distri- if urbanization	ict area consist has an effect o	s of forest ochro of future cocoa	sols (the best c production. Co	ocoa soils). Colur lumn (3): I use t	nn (2): I control for the he share of district area
errors using a distance threshold of 100 km the capital and second largest cities. All the	areau of the 20%. 1. Column (6): e regressions ir 1	I control for the oclude various co	regional value o putrols at the dis	f cash crop exp strict level. The	orts to account see controls are	for spillovers. C described in the	column (7): I dro column (7): I dro e footnote of Tabl	teport coures standard p the national cities, i.e. e 2. See Data Appendix

TABLE 3: URBANIZATION AND NATURAL RESOURCE EXPORTS

					01/T (TO			
Dependent Variable:			Urbanizati	on Rate (%) in year t			Number of Cities in year <i>t</i>
Method:	0LS (1)	OLS (2)	OLS-FE (3)	OLS-FE (4)	OLS-FE (5)	OLS-FE (6)	OLS-FE (7)	OLS-FE (8)
Value of Cash Crop Exports (Standard Score, Between <i>t</i> -1 and <i>t</i>)	6.06*** (0.88)	5.19*** (0.64)	3.87*** (0.68)	3.11^{***} (0.90)	2.07** (0.90)	2.04** (0.88)	3.32^{***} (1.15)	2.17*** (0.38)
Lagged Dependent Variable (Year t – 1)						-0.10 (0.07)	-0.35^{***} (0.11)	-0.07 (0.17)
Value of Cash Crop Exports (Standard Score, Between t-2 and t-1)							4.20** [1.708]	3.99^{***} (1.05)
Year Fixed Effects	γ	Y	Υ	Υ	Y	Υ	Y	A
Baseline Controls x Year	Z	Υ	Υ	Υ	Υ	Υ	Υ	Υ
District Fixed Effects	Z	Z	Υ	Υ	Υ	Υ	Υ	Υ
Region-Year Fixed Effects	Z	Ν	Z	Υ	Υ	Υ	Υ	Υ
District Trends	Z	Ν	Z	Z	Υ	Υ	Υ	Υ
Observations	250	250	250	250	250	250	200	200
R-squared	0.49	0.77	0.82	0.87	0.92	0.92	0.94	0.96
<i>Notes:</i> OLS regressions using panel data for 50 c district urbanization rates over one century. Robt ten regions. The dependent variables in columns (localities > 5,000 inh.) in year t . In all regressic <i>Cash Crop Exports, Between</i> $t - 1$ <i>and</i> t is the star	districts in Ivory ust standard error (1)-(7) is the u (ons, I control fo ndardized distri	Coast in 1948 ors clustered a rbanization ra r the lag of the ct total value of	3.2000. We can t the regional l te (%) in year e dependent vi si t cash crop pr	nnot perform evel are report <i>t</i> for Ivory Cos triable, i.e. the oduction (coco	the same analy ted in parenthe st. In column turbanization a and coffee) l	vsis for Ghana ses; * $p<0.10$ (8) the depen rate or the nul between year i	as I do not ha , ** $p<0.05$, ** dent variable is mber of cities i t - 1 and year	ve panel data on the ** $p<0.01$. There are the number of cities 1 year $t - 1$. Value of t. All the regressions

include various controls at the district level interacted with a time trend. These controls are described in the footnote of Table 2. See Data Appendix for data sources and

construction of variables.

TABLE 4: URBANIZATION AND NATURAL RESOURCE EXPORTS FIXED EFFECTS APPROACH, IVORY COAST, 1948-1998

Appendices

A Data Appendix

This appendix describes in detail the data I use in my analysis.

A.1 Cross-Country Analysis

Spatial Units: I assemble data for 46 Sub-Saharan African countries. I classify them into four regions: Western Africa, Central Africa, Eastern Africa and Southern Africa.

Urbanization Data: I use WUP (2011) to study urbanization rates for 119 developing countries in 2000 and 46 African countries in 1960-2000. The urbanization rate is the share of the urban population in total population (%). There are four different types of urban definition: (i) "administrative": cities are administrative centers of territorial units (e.g., districts, communes, etc.), (ii) "threshold": cities are localities whose population is superior to a population threshold of X inhabitants (e.g., 5,000 or 2,500), (iii) "administrative or threshold": cities are either administrative centers or localities whose population is superior to a population threshold, and (iv) "threshold with condition": cities are localities whose population is superior to a population threshold and whose a large share of the labor force is engaged in non-agricultural activities. We also use WUP (2011) to obtain a list of Asian and African megacities (> 750,000 inh.). WUP (2011) also reports total population for each country in 2000.

Manufacturing and Service GDP and Employment Data: I use WB (2012) to estimate the contribution of manufacturing and services to GDP for 119 developing countries in 2000. Data on the employment structure of the urban sector and largest city in selected countries was recreated using the IPUMS census sample for the closest year to 2000 (IPUMS 2012). Census reports, labor force surveys and household surveys were used for other countries. Table A.1 lists all the countries and years for which we have the urban sectoral composition.

Natural Resource Exports: They consist of fuel, mineral, cash crop and forestry exports. I use WB (2012) and USGS (2012) to estimate the share of fuel and mineral exports in total exports (%) for the 46 African countries every five years in 1960-2000.I use FAO (2012) to obtain the export shares of cash crops and forestry for the same countries and years. Lastly, we use Maddison (2008) and WDI (2012) to obtain the share of merchandise exports in GDP for the same countries and years. Knowing the share of merchandise exports in GDP (%), I can easily reconstruct the contribution of resource exports to GDP (%).

Controls: We use various sources to reconstruct a range of controls at the country-level. Country area (sq km) is obtained from WB (2012). Rural density is defined as the ratio of rural population (1000s) to arable area (sq km) in 2000. The arable area of each country is reported by FAO (2012). Population growth is calculated as the percentage change in country population between 1960 and 2000. We create a dummy if the country is a small island. From wikipedia, we obtain a list of all the island countries in the world. An island country is "small" if its area is smaller than 50,000 sq km. We use GIS to create a dummy if the country is landlocked. We create a dummy if the country has experienced more droughts than the rest of the continent in 1960-2000, 3.8 droughts on average (CRED 2012). We use the Polity IV data series to calculate the average polity score for each country from independence to 2000 (Polity IV 2012a). We then create a dummy if it is lower than -5, the threshold for not being considered as autocratic. Lastly, the Polity IV data series include a measure of political violence for each country from 1964 to date. We create a dummy if the country has ever experienced

an interstate or civil conflict from independence to 2000 (Polity IV 2012b).

A.2 Within-Country Analysis

Spatial Units: I assemble data for a panel of 78 districts in Ghana, from 1891 to 2000, and a panel of 50 districts in Ivory Coast, from 1948 to 1998. Ghanaian districts correspond to *cocoa districts* in 1960, which significantly differ from administrative districts.³⁹ Ivorian districts correspond to administrative districts _ or *départements* _ in 1998.⁴⁰ Ghanaian and Ivorian districts belong to 10 regions in each country.

Urban and Population Data: I collect urban population data from various decadal statistical publications.⁴¹ Defining as a city any locality with more than 5,000 inhabitants, I obtain a geospatialized sample of 364 cities in Ghana (1891, 1901, 1911, 1921, 1931, 1948, 1960, 1970, 1984 and 2000) and 398 cities in Ivory Coast (1901, 1911, 1921, 1931, 1948, 1955, 1965, 1975, 1988, 1998). Using GIS, I can recalculate district urban population for any spatial decomposition, but I am limited by the type of cash crop production data at my disposal. In Ivory Coast, production data is reported at the administrative district level. After reaggregating data to account for administrative changes, I obtain a consistent sample of 50 districts. I only have total population and rural population data from 1948. In Ghana, production is available at the *cocoa district* level (78). As cocoa districts differ from administrative districts, I privilege the former decomposition when creating the urban data set. Besides, administrative boundaries have been considerably modified across years, this impeding any consistent reaggregation. This also means I cannot have panel data on total and rural populations for Ghanaian districts. I nevertheless obtain population data for 36 Southern districts in 1901 and all the districts in 1931.⁴²

Cash Crop Production and Price Data: The data on cash crop production was collected from a variety of sources.⁴³ For both Ghana and Ivory Coast, I am able to create a consistent

⁴⁰The number of Ivorian districts has been increasing over time, but I use various sources and GIS to reconstruct the data set using the same boundaries for the whole period.

⁴¹Publications for Ghana are: *Population and Housing Censuses* 1891, 1901, 1911, 1921, 1931, 1948, 1960, 1970, 1984 and 2000. Publications for Ivory Coast are: *Rapports statistiques* 1900-1920; *Rapports périodiques des gouverneurs et chefs des services* 1895-1940; *Annuaire statistique de l'A.O.F.* 1949-1951 and 1950-1954; *Population de l'A.O.F. par canton et groupe ethnique* 1950-1951; *Répertoire des villages de la Côte d'Ivoire* 1955; *Inventaire économique de la Côte d'Ivoire* 1947-1958; *Côte d'Ivoire* 1965: *Population; Recensement général de la population* 1975; *Population de la Côte d'Ivoire, Analyse des données démographiques disponibles* 1984; *Recensement général de la population et de l'habitat* 1988 and 1998. I also use Geopolis 2010, a previous attempt by geographers to collect urban data for African countries.

⁴²I digitize population maps published in the census reports to obtain population at a fine spatial level in 1901 and 1931. I use GIS to reconstruct population at the district level. As the 1901 census was exhaustive in the South only, I recreate population only for Southern districts.

⁴³Sources for Ghana are: 1927 Yearbook of the Gold Coast; Ghana Population Atlas 1960; Annual Reports and Accounts of the Ghana Marketing Board 1957-1962, 1965, 1970; Dickson (1968); Reports of the Department of Botanical and Agricultural Department 1904-1955; Analysis of Cocoa Purchases by Societies, Districts and Regions 1961-1975, 1989 and 1994-1999; Ghana Cocoa Marketing Board Newsletter 1966-1974; Ghana Cocoa Marketing Board Monthly Progress Reports 1972-1985; and a summary of 2001-2009 district purchases (Ghana Cocoa Marketing Board). Sources for Ivory Coast are: Annuaire Statistique de l'A.O.F. 1949-1951; Inventaire Economique de la Côte d'Ivoire 1947-1958; Documentary Material on Cacao 1947; Annuaire rétrospectif de statistiques agricoles et forestières 1900-1983; Atlas de Côte d'Ivoire, 1971-1979; Caisse de stabilisation et de soutien des prix des productions agricoles; and Pesage systématique du café et du cacao à l'entrée des usines de conditionnement et de transformation 2009.

³⁹Ghanaian production data is not available at the administrative district level. The number of cocoa districts has been decreasing over time, but I use various sources and GIS to reconstruct the data set using the same boundaries for the whole period. As it was not always possible to recreate data by reaggregating or disaggregating districts, I had to make some assumptions when constructing the data. The quality of Ghanaian production data is lower as a result.

dataset of cash crop production (cocoa in Ghana, cocoa and coffee in Ivory Coast) for most years. When a year is missing, it is obtained by linear interpolation, giving production data for 79 districts in Ghana for every year between 1891 and 2009, and 50 districts in Ivory Coast for every year between 1948 and 2009.⁴⁴ I also collect information on regional cocoa yields whenever it is available. I then use the same sources and additional sources to obtain the international and producer prices in year 2000 U.S. dollars.⁴⁵ By multiplying cocoa production going to farmers. Similarly, using the difference between the international and producer price, I get the annual deflated total value of cocoa production being captured by the state. I proceed similarly for coffee in Ivory Coast. Lastly, I estimate the total value of cash crop production between each census year for each district to match the temporal structure of the urban database.

Geographical Data: Forest data comes from land cover GIS data compiled by Globcover (2009). The data displays those areas with virgin forest and areas with mixed virgin forest and croplands, which were areas with virgin forest before it was cleared for cash crop production. I then use historical cocoa soil map which I digitize in GIS to recalculate for each district the area share of each cocoa soil type (forest ochrosols, forest intergrades and forest oxysols).⁴⁶ A district is defined as *suitable* if more than 25% of district area consists of cocoa soils. Suitable land area in the two countries exactly corresponds to the area initially covered by tropical forest. A district is *highly suitable* if more than 50% of district area consists of forest ochrosols, the best cocoa soils. A district is defined as *poorly suitable* if it is suitable, but not highly suitable. I then use GIS: (i) to estimate district area (sq km), (ii) to reconstruct district average annual precipitations (mm) in 1900-1945 and district mean altitude (m),⁴⁷ (iii) to create a dummy for coastal districts, and (iv) to calculate the district Euclidean distances (km) to the coast, the largest city (Abidjan in Ivory Coast, Accra in Ghana).

Mining Production and Price Data: I use annual mining production data for Ghanaian mines in 1891-2000 for four commodities: gold (ounces), bauxite (tons), manganese (tons) and diamond (carats).⁴⁸ As I have the geographical coordinates of each mine and the export price in year 2000 U.S. dollars, I reconstruct the district annual value of mining production.

Household Survey and Census Data: I use household surveys and census microdata to calculate a range of statistics. In Ivory Coast, these are: the 1985-1988 *Living Standards and Measurement Study* (LSMS), the 1998 and 2002 *Enquêtes sur le niveau de vie des ménages* (ENV). In Ghana, these are: the 1987-1988, 1997-1998 and 2005-2006 *Ghana Living Standard Survey* (GLSS), and the 2000 *Population and Housing Census* IPUMS sample.

Agronomic Data: I use various agronomic statistics at the national or regional levels. In addition to household surveys, I exploit Ruf (1995a; 1995b), Teal, Maamah and Zeitlin (2006) and FAO (2012). The aggregate number of cocoa farmers is obtained from census reports and household survey data.

⁴⁴Ghanaian cocoa production data only corresponds to the main crop (October-July). This is not an issue as this amounts to 94.7% of total production from 1948 to 2000.

⁴⁵Additional sources are Bateman (1965), Teal (2002), FAO (2012) and World Bank (2012b). I use parallel exchange rate data when the black market premium is different from zero.

⁴⁶The documents I use for obtaining cocoa soil maps are: *Survey of Ghana: Classification Map of Cocoa Soils* 1958, *Atlas de Côte d'Ivoire: Carte pédologique 1976* and Petithughenin (1998).

⁴⁷Climate data comes from *Terrestrial Air Temperature and Precipitation: 1900-2006 Gridded Monthly Time Series, Version 1.01.* Altitude data comes from NASA/JPL SRTM30.

⁴⁸Mining production and price data is collected from the following documents: *The Mineral Industry of the British Empire and Foreign Countries 1913-1919; Reports of the Mines Department of the Gold Coast 1931-1958;* and *The Mineral Industry of Ghana 1963-2000 (USGS 2012).*

Infrastructure Data: I collect data from various sources on the spatial allocation of infrastructure. First, I create three control variables intended to capture transportation infrastructure at independence (1958 in Ghana, 1960 in Ivory Coast). A GIS data set of railways and roads in 2010 is obtained from Digital Chart of the World. I then use various sources to identify international ports and reconstruct the railway network around 1960: Dickson (1968) and Atlas de Côte d'Ivoire, 1971-1979. I use Michelin road maps to reconstruct in GIS the road network around 1960. Michelin road maps allow me to distinguish paved and unpaved roads. I then use GIS to create district dummies for whether a district is connected to the railway network in 1960, or the paved road network in 1960, or contains an international port in 1960. Second, I collect infrastructure data for a cross-section of districts in 2000. For Ghana, I use the 2000 Facility Census to calculate for each district the share of rural and urban inhabitants less than 10 km from the following facilities: primary school, junior secondary school, senior secondary school, hospital, health centre, post office and telephone. I then use the 2000 Population and Housing Census to calculate for each district the share of rural and urban inhabitants with access to electricity and tapwater. Lastly, I use a recent Michelin road map which I have digitized in GIS to estimate the district density of paved roads (in meters per sq km).

Demographic Data: Using various sources, I am able to track the evolution of birth and death rates separately for urban and rural 1960-2000 Ghana and 1965-1998 Ivory Coast.⁴⁹

REFERENCES

Bateman, Merrill. 1965. *Cocoa in the Ghanaian Economy*. Unpublished Ph.D. dissertation, Massachussets Institute of Technology.

CRED. 2012. The OFDA/CRED International Disaster Database. Louvain: CRED, Université catholique de Louvain.

Dickson, Kwamina. 1968. A Historical Geography of Ghana. Cambridge: Cam- bridge University Press.

FAO. 2012. FAOSTAT. Roma: Food and Agriculture Organization.

Globcover. 2009. Globcover Africa. Louvain: European Space Agency Ionia.

ILO. 2012. LABORSTA. Roma: International Labor Organization.

IPUMS. 2012. Integrated Public Use Microdata Series, International. Minneapolis, MN: Minnesota Population Center.

Maddison, Angus. 2008. *Historical Statistics of the World Economy: 1-2008 AD*. Groningen Growth and Development Center.

Petithuguenin, Philippe. 1998. "Les conditions naturelles de production du cacao en Cote d'Ivoire, au Ghana et en Indonesie." *Plantations, Recherche, Developpement*, 5(6): 393-411.

Polity IV. 2012a. Polity IV Project: Political Regime Characteristics and Transitions, 1800-2010: Polity IV Annual Time-Series 1800-2011. Vienna: Center for Systemic Peace.

Polity IV. 2012b. Polity IV Project: Political Regime Characteristics and Transitions, 1800-2010: Major Episodes of Political Violence, 1946-2008. Vienna: Center for Systemic Peace.

⁴⁹The main sources for Ghana are: *The Population of Ghana 1974*; *Demographic and Household Survey* 1988; and *Ghana's Development Agenda and Population Growth* 2006. The main sources for Ivory Coast are: Chaleard (2000); Tapinos, Hugon and Vimard (2003); and *Recensement général de la population et de l'habitation 1998*.

Ruf, Francois. 1995a. *Booms et crises du cacao: Les vertiges de l'or brun*. Paris: Ministere de la Coopeeration, CIRAD-SAR and Karthala.

Ruf, Francois. 1995b. "From "Forest Rent" to "Tree Capital": Basic Laws of Cocoa Supply." In The Economics of Cocoa Supply. , ed. F. Ruf and P S Siswputranto. Cambridge: Woodhead Publishing Limited.

Teal, Francis J. 2002. "Export Growth and Trade Policy in Ghana in the Twen- tieth Century." *The World Economy*, 25(9): 1319-1337.

Teal, Francis J., Andrew Zeitlin, and Haruna Maamah. 2006. Ghana Cocoa Farmers Survey 2004: Report to Ghana Cocoa Board. CSAE-Oxford University.

USGS. 2012. *Bureau of Mines Minerals Yearbook 1963-2010*. Reston: United States Geological Survey.

World Bank. 2012a. International Financial Statistics. Washington D.C.: World Bank.

World Bank. 2012b. World Development Indicators. Washington D.C.: World Bank.

WUP. 2011. World Urbanization Prospects, the 2011 Revision. New York: United Nations.

B Model Appendix: Natural Resource Exports and Urbanization

In this appendix I develop a model where resource exports drive urbanization. To highlight the mechanism as clearly as possible, the model assumes a small economy, one production factor - labor - and four sectors: food, resources, urban tradables and urban non-tradables.

B.1 Set-Up

There are four sectors *i*: food (*f*), natural resources (*r*), urban tradables (*d*) and urban non-tradables (*n*). Food and resources are tradable. Urban tradables are tradable manufactured goods and services. Urban non-tradables are non-tradable goods and services.⁵⁰ There is one representative agent endowed with one unit of labor. The production technologies are:

$$Y_i = A_i L_i. (2)$$

where Y_i , A_i and L_i are the output, productivity and labor share of sector *i*. Productivity is exogenous. Food and resources are produced in the non-urban sector. Manufactured goods and services being produced in urban areas, the urbanization rate is $L_u = L_d + L_n$. The representative agent has the following non-homothetic preferences:

$$U(C_f, C_r, C_d, C_n) = (C_f - \bar{C_f})^{\alpha_f} (C_r)^{\alpha_r} (C_d)^{\alpha_d} (C_n)^{\alpha_n}.$$
(3)

where C_f , C_r , C_d and C_n denote the consumption of food, resources, urban tradables and urban non-tradables. Resources are not used as intermediary goods in other sectors. The sum of consumption weights is equal to one ($\Sigma_i \alpha_i = 1$). \bar{C}_f is a food subsistence requirement. With $\bar{C}_f > 0$, the income elasticity of the demand for food is below one. The representative agent maximizes utility (2) subject to the budget constraint:

$$w = p_f C_f + p_r C_r + p_d C_d + p_n C_n.$$
(4)

where w, p_f , p_r , p_d and p_n are the wage rate and the prices for each sector.

B.2 A Closed Subsistence Economy

I first consider a closed economy. The consumer maximizes utility subject to the budget constraint. This yields the following demands:

$$C_f = \bar{C}_f + \frac{\alpha_f}{p_f} (w - p_f \bar{C}_f), \qquad C_j = \frac{\alpha_j}{p_j} (w - p_f \bar{C}_f) \text{ for } j = \{r, d, n\}.$$
(5)

The consumption of *i* is increasing in the wage rate *w* and *i*'s utility weight α_i , and decreasing in *i*'s price p_i . The food subsistence requirement \bar{C}_f increases food consumption C_f but decreases the consumption of other goods. Perfect competition for labor implies that:

$$w = p_f A_f = p_r A_r = p_d A_d = p_n A_n.$$
(6)

Goods and labor markets clearing conditions are:

$$C_i = Y_i = A_i L_i , \quad \Sigma_i L_i = 1.$$
⁽⁷⁾

⁵⁰This sector addresses the specific demand for "domestic" manufactured goods and services.

Combining (4)-(6), we find the following demands and labor shares:

$$C_f = \alpha_f A_f + (1 - \alpha_f) \bar{C}_f$$
, $C_j = \alpha_j A_j - \alpha_j \frac{A_j}{A_f} \bar{C}_f$ for $j = \{r, d, n\}$, (8)

$$L_f = \alpha_f + (1 - \alpha_f) \frac{\bar{C_f}}{A_f}, \quad L_j = \alpha_j - \alpha_j \frac{\bar{C_f}}{A_f} \quad \text{for } j = \{r, d, n\}.$$
 (9)

The food labor share is greater than α_f but converges to it as food productivity increases. Conversely, the labor share of other sectors is lower than the consumption weight, but converges to it as the food constraint is relaxed. The representative agent lives at the subsistence level if food productivity is low enough. Assuming $A_f = \bar{C}_f$, market clearing gives:

$$C_f = Y_f = A_f L_f = \bar{C_f} L_f.$$
⁽¹⁰⁾

Consumption C_f is below the subsistence requirement \overline{C}_f unless $L_f = 1$. When food productivity is sufficiently low, she only consumes food and only works in the food sector:

$$C_f = \bar{C_f}, \quad C_j = 0 \quad \text{for } j = \{r, d, n\}, \quad L_f = 1, \quad L_j = 0 \quad \text{for } j = \{r, d, n\}.$$
 (11)

The movement of labor out of the food sector into other sectors is constrained by the need to satisfy food requirements. There cannot be urbanization without a food surplus.

B.3 Natural Resource Exports in a Small Open Economy

I now consider a *small open economy*. The agent takes international prices p_f^* , p_r^* and p_{ut}^* as given. If the economy has a comparative advantage in natural resources, the autarky relative price of resources is lower than the international relative price:

$$p_r/p_f < p_r^*/p_f^*, \quad p_r/p_d < p_r^*/p_d^*.$$
 (12)

The structure of comparative advantages implies that the country exports resources (X_r) and imports its consumption of food (M_f) and urban tradables (M_d) . Demands are not changed, but the country has two producing sectors and perfect competition for labor means that:

$$w = p_r^* A_r = p_n A_n. \tag{13}$$

Goods and labor markets clearing conditions are:

$$X_r + C_r = Y_r = A_r L_r$$
, $C_n = Y_n = A_n L_n$, $C_f = M_f$, $C_d = M_d$, $\Sigma_i L_i = 1$. (14)

The balanced trade assumption stipulates that imports equal exports:

$$p_r^* X_r = p_f^* M_f + p_d^* M_d.$$
(15)

Using (4) and (12), and defining $R = p_r^* A_r - p_f^* \overline{C}_f$ as the surplus income available to individuals once they have met their subsistence requirement, we find the following demands:

$$C_{f} = \bar{C_{f}} + \alpha_{f} \frac{R}{p_{f}^{*}}, \quad C_{j} = \alpha_{j} \frac{R}{p_{j}^{*}} \quad \text{for } j = \{r, d\}, \quad C_{n} = \alpha_{n} A_{n} \frac{R}{p_{r}^{*} A_{r}}.$$
 (16)

Since $\bar{C_f} = A_f$ and $p_r/p_f < p_r^*/p_f^*$, R > 0 and $C_f > \bar{C_f}$. The country gains from trade. The share of available income *R* allocated to sector *i* increases with *i*'s consumption weight α_i and

decreases with *i*'s price p_i . Combining (14)-(17) and (18), the labor shares are:

$$L_{r} = (1 - \alpha_{n}) + \alpha_{n} \frac{p_{f}^{*} \bar{C}_{f}}{p_{r}^{*} A_{r}}, \quad L_{n} = \alpha_{n} - \alpha_{n} \frac{p_{f}^{*} \bar{C}_{f}}{p_{r}^{*} A_{r}}.$$
 (17)

First, the country urbanizes, because the food constraint has been relaxed and the consumption of urban non-tradables has increased. Now, any rise in the resource rent $(p_r^*A_r)$ increases the labor share of urban non-tradables. Assume there are two types of resource sectors, fuels and mining where productivity A_r is high and cash crops where it is much lower. The urbanization rate is higher in the former, because the surplus per worker is higher. In the extreme, resource productivity is so high (like a "manna from heaven") that the country fully urbanizes. Second, urbanization takes place in consumption cities with non-tradable workers only.

B.4 Discussion and Possible Extensions

The model shows that resource exports can produce urbanization without structural transformation. I now discuss how the model could me modified to become more realistic. See Gollin, Jedwab & Vollrath (2013) for a complete model of resource exports and urbanization.

Complete specialization. First, the current model is overly simplistic as it predicts that the food sector disappears. The model could be enriched by modeling imperfect substitutability between domestic and foreign foodstuffs and/or allowing for internal trade costs that would make food imports too expensive for the hinterland. Second, the country does not industrialize at all. However, if there is imperfect substitutability and/or internal trade costs, there could be some employment in the urban tradables sector. We could also permit capital accumulation in that sector, which could increase productivity over time and change the country's comparative advantage in the longer run. It is perfectly possible that the country experiences structural change ex post. However, if there are dynamic externalities in the urban tradables sector, the original structural of comparative advantages will be reinforced over time, and it will be difficult for the country to compete with countries that industrialized earlier.

The sectoral composition of consumption cities. First, the export of natural resources and the import and domestic trade of food and urban tradables could create employment in the trade and transportation sectors. The labor share of urban non-tradables could be a function of the trade volume: $L_n = A_n Y_n = A_n [C_n + \zeta (X_r + M_f + M_d)]$. Second, the government could tax the resource sector: $w = (1 - \tau)p_r^*A_r$, with τ being the tax rate. This augments the effect of natural resources on urbanization if the government's Engel curve for urban non-tradables (e.g. civil servants) is steeper than the private Engel curve $(\alpha_n^g > \alpha_n)$. If there are two regions - *production* and *government* - and the government concentrates its urban consumption in its own region, this increases the primacy rate. Government cities are also consumption cities. Lastly, if some resource workers have a strong preference for urban living (i.e., a high α_n), the wealthiest of them will live in city and these agro-towns are also consumption cities.

Urbanization without urban tradables and long-run growth. There are several channels through which this type of urbanization could have a relatively lower effect on long-run growth. First, there could be dynamic externalities in the urban tradables sector or incentives for productivity improvement could be greater due to the threat of global competitors: $\Delta A_d = \lambda A_d$, with $\lambda > 0$. Second, there could be large intersectoral linkages from that sector. Third, that sector could provide more incentives for human capital accumulation, if the production technology is $Y_d = A_d L_d H_d$ and H affects productivity in the other sectors: $\Delta A_{-d} = \eta H_d$, with $\eta > 0$. Fourth, resource exports could have a negative effect on the quality of institutions θ , that would affect sectoral productivities: $A_i = f_+(\theta)$, for all i.

Country	Year	Largest City	Source
Angola	1993	Luanda	Labor Force Survey 1993 (ILO)
Benin	2002	Cotonou	Questionnaire des Indicateurs de Base du Bien-Etre 2002
Botswana	2005	Gaborone	Labor Force Survey 2005 (ILO)
Burkina Faso	1998	Ouagadougou	Enquete Prioritaire 1998
Ethiopia	2004	Addis Ababa	Labor Force Survey 2004 (ILO)
Gambia	1993	Banjul	Household Economic Survey 1993
Ghana	2000	Accra	Population and Housing Census 2000 (IPUMS)
Guinea	1996	Conakry	Population and Housing Census 1996 (IPUMS)
Guinea-Bissau	1993	Bissau	Inquerito ao Consumo e Orcamentos Familiares 1993
Ivory Coast	2002	Abidjan	Living Standard Survey 2002
Kenya	1999	Nairobi	Labor Force Survey 1999 (ILO), Population and Housing Census 1989 (IPUMS)
Liberia	2010	Monrovia	Labor Force Survey 2010 (ILO)
Madagascar	2002	Antananarivo	Enquete Aupres des Menages 2001, 2002, Population and Housing Census 1993 (Census Report)
Malawi	1998	Lilongwe	Population and Housing Census 1998 (IPUMS)
Mali	1998	Bamako	Population and Housing Census 1998 (IPUMS)
Mauritius	2000	Port-Louis	Population and Housing Census 2000 (Census Report)
Mozambique	2004	Maputo	Labor Force Survey 2004 (ILO)
Namibia	2000	Windhoek	Labor Force Survey 2000 (ILO)
Nigeria	2005	Lagos	General Household Survey 2005
Rwanda	2002	Kigali	Population and Housing Census 2002 (IPUMS)
Senegal	2004	Dakar	Enquete Senegalese Aupres des Menages 2004
Sierra Leone	2004	Freetown	Population and Housing Census 2004 (IPUMS)
South Africa	2001	Johannesburg	Population and Housing Census 2001 (IPUMS)
Sudan	2008	Khartoum	Population and Housing Census 2008 (IPUMS)
Tanzania	2006	Dar Es Salaam	Labor Force Survey 2006 (ILO)
Uganda	2002	Kampala	Population and Housing Census 2002 (IPUMS)
Zambia	2008	Lusaka	Labor Force Survey 2008 (ILO)

TABLE A.1: URBAN WORKFORCE SOURCE INFORMATION BY COUNTRY

Notes: Employment data are for all economically active persons aged 15+. IPUMS is the International Public-Use Microdata Series. ILO is the International Labor Organization. Census reports and household surveys were used for other countries. These 27 countries account for 80.0% of Sub-Saharan Africa's population in 2000. The large African countries (> 10 million inh. in 2000) for which no data could be found are the Democratic Republic of the Congo, Cameroon, Zimbabwe and Niger.